DESIGN OF A PHASE LOCKED LOOP BASED CLOCKING CIRCUIT FOR HIGH SPEED SERIAL LINK APPLICATIONS

BY

RISHI RATAN

THESIS

Submitted in partial fulfillment of the requirements for the degree of Master of Science in Electrical and Computer Engineering in the Graduate College of the University of Illinois at Urbana-Champaign, 2014

Urbana, Illinois

Adviser:

Professor José Schutt-Ainé

ABSTRACT

Technology scaling and unprecedented growth in demand for ubiquitous, fast, robust computing have been the driving forces leading the innovations in high-speed interfaces. With the rise of heavy duty data centers to handheld mobile devices, the desire for faster, low-power integrated inter-IC communication protocols is at an all-time high and has led the roadmap of the semiconductor industry, making it one of the fastest growing yet fiercely competitive industries. With the growing needs for ultra-low power yet multi-Gbps signaling in both wired as well as wireline applications, integrated systems on chip (SoCs) have become mainstream critical components in modern computing systems. The ability to process and access 'big-data' is the fundamental demand in modern society where every second saved in prompt communication as well as computation of information is critical. In order to meet these needs of fast, robust signaling over the same old "lossy" channels, the clock-frequencies need to scale accordingly and clever I/O links need to be developed. The most crucial component of any high-speed I/O link is the clocking circuitry: clock generator at the transmit (TX) end and clock-recovery unit on the receive (RX) end.

This thesis provides an in-depth tutorial on circuit design, analysis and simulation of on-chip PLL based clocking generator circuits for high-speed serial link applications. An overview of high-speed links, along with the basic building blocks that make up a serial link, is presented. The fundamentals of PLLs are introduced and a complete guide to analysis and simulation of a charge-pump phase-locked loop based clocking circuit at both behavioral as well as transistor levels is presented for use as a synthesizer in a serial link. Finally, a survey of potential future research areas to explore for both PLLs in high-speed links as well as the complete serial link is provided with an emphasis on signal integrity applications for future students pursuing graduate studies in the fields of Signal Integrity and Mixed-Signal IC Design. To my family and friends, for their love and support.

ACKNOWLEDGMENTS

As I approach the finishing stages of my graduate career, looking back there are countless individuals who have helped make this journey special and memorable. Graduate school is full of numerous uncertainties, various roadblocks in terms of design and implementation of ideas; thus it is a journey that though embarked on by one individual, really is a culmination of the efforts of many people who have helped behind the scenes along the way. At the end of it all, it is this support system that I am deeply indebted to for helping each step of the way, motivating me in the most difficult of times where there were only questions and no answers.

First and foremost, I want to thank my wonderful advisor Professor José Schutt-Ainé for being my mentor, always motivating me to push myself to the next level and being there for me like family. Dr. Schutt-Ainé was one of the first to see potential in me back when I was just a sophomore and had no direction in what I wanted to pursue in my career. He introduced me to the field of circuits and sparked an interest in engineering. He has always been extremely patient throughout the countless occasions I have visited his office hours. Being a mentor and father-figure for me throughout my educational career at UIUC, Prof. Schutt-Ainé has been the biggest contributor to my successes throughout my collegiate career and will forever be a role-model who I look up to as I start my professional career in industry.

Secondly, I would like to convey my heartfelt thank you to Professor Pavan Kumar Hanumolu for guiding me throughout this thesis as a mentor. Prof. Hanumolu stood by me with patience throughout the course of this thesis work, and took time off from his busy schedule in providing me formal lectures on PLLs one-on-one in his office at a time when it was very hectic for him given that he had just moved to UIUC. Without Dr. Hanumolu's constant guidance and feedback this thesis would never have been completed.

Thirdly, I would like to sincerely thank my wonderful colleagues, mentors

and friends in graduate school: Da Wei, Xu Chen, Tom Comberiate, Romesh Nandwana, Mrunmay Talegaonkar, Yubo Liu, Ahmed Elkholy, Saurabh Saxena, Tejasvi Anand, Guanghua Shu, Woo-Seok Choi, Drew Newell, Xinying Wang, Jerry Yang and Maryam Hajimiri for always being patient to answer all my questions, solving arcane problems during the simulation and design process. These fine graduate students have stood by me since day one, provided constructive criticism on my work, painstakingly critiqued every figure and most importantly constantly pushed me towards striving for nothing short of excellence.

Furthermore, I am extremely grateful to Professor Steven Franke, Professor Christopher Schmitz, Professor Milton Feng, Professor Elyse Rosenbaum, Professor Naresh Shanbhag, Professor Venugopal Veeravalli and Dr. Chandrashekhar Radhakrishnan for supporting me throughout my educational career at UIUC and mentoring me every step of the way. Even when the chips were down, they always believed in me and helped me gain an opportunity to pursue my graduate studies.

I thank my family for always being by my side, motivating me to move forward whenever I was faced with roadblocks. Lastly, I am eternally grateful to my friends Ian Wetherbee, Rohan Bambery, Eric Iverson, Pourya Assem, Sai Zhang, Min-Sun Keel, Anish Chivukula, Eclair Hanjing Gao, Dennis Yuan, Jerry Sun, Nishant Nookala and Eric Kim for always standing by me throughout my ECE career at UIUC. I am positive I have missed a few people from the above list but I truly am very thankful to all those who I have interacted with throughout my time at UIUC. Additionally, I am very grateful for getting an opportunity to work with wonderful undergraduates Rushabh Mehta, Ishita Bisht, Brady Salz, Ankit Jain, Haodong Guo, and Pradyut Paul who have worked very diligently in getting the High Speed-SerDes Design project off the ground from scratch.

TABLE OF CONTENTS

CHAPT 1.1 1.2	TER 1 INTRODUCTION INTRODUCTION Motivation Intervention Outline Intervention	1 1 3
CHAPT 2.1 2.2 2.3	TER 2 HIGH SPEED SERIAL LINKS OVERVIEW . . Simple Link Design . . . Serial vs. Parallel Data Transmission . . . SerDes Building Blocks . . .	5 5 7 8
CHAPT 3.1 3.2	TER 3 PLL THEORY AND BACKGROUND	17 17 17
CHAPT 4.1 4.2 4.3	TER 4PLLs IN CLOCKING CIRCUITSCharge-Pump (CP)PLLs OverviewCPLL Linear Model and AnalysisCPLL Noise-Analysis	23 23 24 26
$\begin{array}{c} \text{CHAPT} \\ 5.1 \\ 5.2 \\ 5.3 \\ 5.4 \\ 5.5 \end{array}$	ER 5 PLL BASED CLOCK GENERATOR	29 29 31 32 33 34
CHAPT 6.1 6.2 6.3 6.4	TER 6BEHAVIORAL LEVEL SIMULATIONWhy Behavioral Modeling?Why Verilog-AMS?Basic Verilog-A/AMS SyntaxPLL Simulation in AMS Using Cadence Virtuoso	35 35 35 36 38
CHAPT 7.1 7.2 7.3 7.4	TRANSISTOR LEVEL SIMULATION What is SPICE? SPICE vs. Spectre Transient, PSS and PNoise Simulation Overview PLL Simulation in Spectre Using Cadence Virtuoso	54 54 54 55 56

CHAPTER 8 DISCUSSION		
8.1 Conclusion		
8.2 Future Work		
APPENDIX A CADENCE VIRTUOSO INSTALLATION GUIDE 92		
A.1 Introduction $\dots \dots \dots$		
A.2 Environment Setup		
A.3 Common Troubleshooting Tips		
REFERENCES		

CHAPTER 1 INTRODUCTION

1.1 Motivation

Over the last 50 years, advances in Semiconductor Fabrication Technology (SFT) coupled with innovations in Integrated Circuit (IC) technology scaling have fueled an unparalleled growth in computing. This aggressive scaling has revolutionized every aspect of modern society and triggered an insatiable demand for faster data rates and higher processing power resulting in clock frequencies and corresponding data rates approaching multi-GHz and multi-Gbps ranges in everyday computing devices like personal computers, mobile devices, entertainment consoles and other such devices. Access to information promptly and efficiently in terms of power and portability/ease of use is the major driver pushing the limits of IC technology. Thus, the need for robust, high-speed, low-power and highly integrable compact systemson-chip (SOCs) is paramount for inter-IC communication interfaces such as network switches, processor/memory interfaces across backplane channels. In order to meet this growing demand for wideband systems, the Input/Output (I/O) links need to scale proportionally with the increased data-rate scaling; however in reality the off-chip I/O bandwidth (BW) has not scaled appropriately and has become a major bottleneck in the overall system performance. Furthermore, along with the off-chip I/O BW limitations, the channel as well as package/connector interfaces have not scaled with SFT making the design of high-speed I/O links extremely challenging due to the increased transmission line loss, crosstalk, and signal distortion resulting in intersymbol interference. As the demand for high data-rate interfaces has skyrocketed, the clock-frequencies needed to realize such systems have correspondingly reached the multi-GHz range necessitating the use of phase-locked loops (PLLs) for on-chip clock synthesis.



Figure 1.1: IO Link Signaling Data-Rate Trends

Figure 1.1 shows the trends in data-rate scaling of I/O high-speed signaling links as forecasted by the International Solid State Circuits Conference (ISSCC) 2011 annual semiconductor roadmapping report [5]. The key takeaway from this graph is that the data-rates in inter-IC communication links are scaling by a factor of 2X every 4 calendar years while IO channel BW remains the same. The I/O BW scaling problem aside, the ability to design robust, low-jitter on-chip clock synthesizer circuits is in itself an extremely challenging task. Though research in the field of integrated high-frequency clocking circuits has been going on for the past two-decades and lots of innovative designs have come into existence, one common facet missing from the whole paradigm is complete documentation on process of simulation using the Electronic Design Automation (EDA) tools [6]. Most of the literary works in this area primarily focus on novel system level designs for PLL based clock synthesizers and some go into transistor-level details of the sub-blocks; however very rarely do any of the prominent works describe the actual simulation process. As the clock-frequencies scale and demand for robustness in the on-chip synthesizers increases, circuit designers also need to be aware of potential Signal-Integrity (SI) problems associated with their intricate designs. Since the channel BW is essentially the same at high-frequencies of operation the PCB traces act as transmission-lines (TLs) leading to severe degradation in signal quality due to reflections, ringing and cross-talk effects. Thus, every integrated circuit designer will invariably face SI problems in their design which until recently was not a concern as the frequencies were low enough that digital design did not require a formal understanding of signal integrity during the development as well as verification process. Therefore, the motivation for this thesis is to bridge the gap between circuit design and simulation for signal-integrity engineers who need the basic expertise in mixed-signal design process to be able to provide the required assistance to IC designers on designing high-speed SI aware systems.

1.2 Outline

This thesis is organized to serve as a training manual for students pursuing mixed-signal integrated circuit design as their field of study in graduate school. The goal is for this thesis to be their go-to guide to grasp a high-level understanding of high-speed links and learn the simulation setup/procedure to validate PLL based clocking circuits using the popular EDA tool Cadence Virtuoso.

- 1. Chapter 1 provides an introduction to the research problem describing the need for high-speed serial links and their future trends.
- 2. Chapter 2 provides an overview of high-speed links with an emphasis on describing each of the building blocks, figures of merit to characterize these links and lay the motivation for the industry-wide shift from parallel to serial-link design for low power, cost-effective robust I/O link design.
- 3. Chapter 3 describes the fundamentals of Phase-Locked Loops (PLLs) and provides a brief overview of their ubiquitous use in modern day wireline/wireless systems.
- 4. Chapter 4 covers a special class of PLLs, Charge-Pump PLLs, and provides a linear model for small-signal as well as noise-analysis of these PLLs.
- Chapter 5 presents the transistor-level design of a Charge-Pump based Integer-N clock generator circuit operating at an output frequency of 1.6GHz.

- Chapter 6 describes the procedure for behavioral modeling and simulation using Verilog-AMS for the clock-generator circuit described in Chapter 5.
- 7. Chapter 7 describes the procedure for transistor-level simulation for the clock-generator circuit described in Chapter 5.
- 8. Chapter 8 concludes the thesis with a discussion of the take-aways from the clocking circuit designed earlier and provides a brief anecdote on the signal integrity focus areas in high-speed link design and lists design improvements on the basic Integer-N analog clock generating circuit to accommodate industry trends within the field.
- 9. Lastly, the Appendix provides a step-by-step guide for installing and configuring the Cadence Virtuoso environment.

CHAPTER 2

HIGH SPEED SERIAL LINKS OVERVIEW

2.1 Simple Link Design



Figure 2.1: Typical High-Speed Link Block Diagram

Generalized model of a High-Speed Serial Link (HSSL), as shown in Figure 2.1 [1], consists of a serializer and transmitter (TX) driven by a PLL clock synthesizer, a channel and a receiver (RX) and Deserializer driven by a Clock-Data Recovery (CDR) unit. The serializer accepts the incoming parallel data-stream and converts it into a serial data-stream which is then sent to the transmitter. The TX generates a train of pulses depending on the data symbols to be transmitted across the channel and the pulse-width which is determined by the timing instant of the transmit clock at both begin/end/edges. The receiver basically comprises a sampler and a decision circuit whose purpose is to sample the received data-bit stream from the channel and recover, both the transmitted data as well as the clock. Once the receiver recovers the transmitted serial bit-stream it is sent to the Deserializer block whose job as the name suggests is to convert the received serial data back to its original parallel form for future interfaces.



Figure 2.2: Typical Backplane Channel Interface

As discussed in the previous chapter, though the desired data-rates have scaled along with technology scaling, the off-chip channel I/O BW has remained the same. The channel is the electrical path between the TX and RX blocks and in inter-IC communication systems typically comprises printed circuit-board (PCB) traces, vias, connectors and other such I/O interface components. Generally speaking, in high-speed I/O interfaces the channel is typically a 'backplane' which essentially connects two PCBs together and typically looks like the interface shown in Figure 2.2 [1]. The channel is something that the designer has no control over and is thus just a 'known unknown' to the link designer. It is known in the sense that the channel impulse-response is known via measurement of S-parameters of the interface using a Vector-Network-Analyzer (VNA) or via computational electromagnetic modeling software such as Ansys HFSS. The manner in which channel degrades the transmitted signal stream is the unknown aspect and the designer's aim is to design a mechanism for counteracting this degradation. Thus, the whole challenge in high-speed link design is that we need to design a high signal-fidelity communication system that is fast, robust to losses incurred in the channel and on top of all this it needs to be low-power and must occupy the least possible area. The challenge in meeting all of the aforementioned requirements is that at high-speeds the channel suffers from various kinds of microwave losses due to impedance discontinuities between connectors, substrate loss, cross-talk effects, reflections and ringing, all which are difficult to predict and model [1].

Figure 2.3 [2] shows an example of the I/O link interface for a 10Gbps serial link across a backplane channel. Notice that a clean signal when transmitted across a backplane channel incurs tremendous amounts of loss from the channel at an operating speed of 10Gbps making the signal at the receiver virtually indistinguishable from noise and thus virtually garbage. In order to limit the degradation of signal quality during transmission and reception, the goal of mixed-signal designers is to design fast high-frequency clocks with minimum timing skew at the TX end and minimal sampling errors at the RX end.



Figure 2.3: 10Gbps Backplane Serial Link Interface

2.2 Serial vs. Parallel Data Transmission

Historically, parallel links have been widely used in I/O systems that are connected to the CPU in computers via interfaces like PCI, PCI-X buses. However, as the data-rates have scaled into multi-gigabit ranges, the parallel link performance has not scaled accordingly with high signal fidelity. The tolerance level in timing skew between parallel signaling links has reached the practical limit achievable using traditional printed-circuit-boards (PCBs) that typically use FR-4 substrates. Additionally, as the supply voltage levels in modern CMOS process technologies have scaled down tremendously, the legacy parallel bus voltage levels have not scaled proportionally, making them incompatible with modern processes [6]. Thus, in order to mitigate this performance limitation and supply voltage scaling problems posed by conventional parallel-link design the industry has shifted to electrical point-to-point serial link interfaces.

Serial links occupy small area on chip and require very few I/O pins as compared to case of parallel links because the number of pins is not directly proportional to the number of data input/output signals. In serial communication links clock-skew is not a problem at the receiver since TX clock is typically not forwarded to the RX. In parallel links, on the other hand clockskew is the major source of signal degradation at the RX side since the TX clock and data are transmitted separately. Furthermore, cross-talk effects are minimized in serial links due to the absence of multiple conducting channels in parallel that each have varying signals transmitted, whereas in parallel links this is a major problem due to the presence of capacitive/inductive coupling between multiple conducting parallel interconnect channels.

In the consumer electronics industry, serial links have found widespread acceptance in the form of USB (Universal Serial Bus) that connects peripheral electronic systems to computer, and SATA (Serial Advanced Technology Attachment) which connects the computer motherboard with mass storage devices (e.g. hard disk) and PCI-Express (Peripheral Component Interconnect) that is used to connect cards (sound, video or other) to the motherboard. Therefore serial communication has become the solution to higher and more efficient data transmission in order to meet the demands and trends of the higher capacity of communication technology [7].

2.3 SerDes Building Blocks

2.3.1 Serializer

The serializer circuit, as the name suggests, converts the input parallel-bus data into a serial bit-stream form. It is a completely digital block and it precedes the TX driver circuit. At a fundamental level, a serializer is essentially a Multiplexer circuit whose driving clock for the serialization process is the TX_CLK signal generated by the TX PLL.

2.3.2 Driver Amplifier

Driver amplifiers are found both at the TX as well as RX ends. The DA (Driver Amplifier) is used to amplify the input serial bit-stream before it is sent to the receiver through the channel. Another important task accomplished by the DA is that it provides impedance terminations that terminate the channel input/output with 50Ω impedance.

2.3.3 Phase-Locked Loop (PLL) Clock Generator



Figure 2.4: Typical PLL Based Clock-Generator Block Diagram

A PLL is a negative feedback system whose sole purpose is to use a reference input clock of frequency f_{REF} and generate a local clock on-chip at a desired frequency, f_{OUT} , such that the output clock is matched in phase to the input clock and $f_{OUT} = \alpha f_{REF}$, where α is a multiplying factor. PLLs are used in every modern day high-speed system whether it be wired or wireless because generating a high spectral purity clock at microwave frequencies/data-rates is practically not feasible yet. Piezo-electric crystals are used exclusively as the reference clocks for almost every on-chip interface system because they have the highest-spectral purity and can output truly periodic, jitterfree clock signals typically up to 200MHz. Due to the insatiable demand for robust, high-speed signaling, a mere crystal oscillator is not enough to meet the necessary requirements, so a PLL is essential. The most important task of a PLL is therefore to produce clock signals with minimal timing noise, i.e. the lowest possible jitter (in time domain) and phase-noise (in frequency-domain). At a block-level, a typical PLL clock-generator circuit (as shown in Figure 2.4) used to generate the high-speed on-chip clocks consists of a phase-frequency detector (PFD), charge-pump (CP), loop-filter (LF), voltage-controlled oscillator (VCO) and clock-divider (DIV). PFD tracks the phase and frequency difference between the reference signal and the divider

output signal outputs digital pulse-width modulated (PWM) signals to the CP which essentially converts these digital pulses into an analog current signal. The LF then takes the CP output current signal, low-pass filters the high-frequency noise components and outputs a control voltage that drives the VCO. The VCO is the most-critical component within the PLL as it is the circuit-block that generates the final output clock that is used to drive the digital circuits of the link. Thus, a low phase-noise VCO is of paramount importance in the PLL as the VCO phase-noise is the dominant noise-form in the PLL. Finally, the divider is used in the feedback loop back to the PFD as the VCO output needs to be brought back down to the same frequency level as the reference clock so that the loop can dynamically drive all static-phase errors between reference clock and divider clock to zero such that $f_{OUT} = \alpha f_{REF}$, where α is the multiplying factor and the loop is "locked" to output a stable clock at the desired frequency of operation. The various intricacies involved in PLL design are covered in the remainder of the thesis.

2.3.4 Channel



Figure 2.5: Typical Serial Link Channel Responses

Figure 2.5 [3] shows the attenuation levels a typical serial link channel incurs as a function of operating frequency. Figure 2.6 and 2.7 show the eye diagram outputs from a backplane channel interface at 1Gbps and 10Gbps data-rates respectively. Notice that the eye is fully open at 1Gbps but at 10Gbps the signal is almost indistinguishable from the noise at the receiver side due to the tremendous loss and distortion incurred along the channel. The HSSL designers need to be able to account for such losses when designing the blocks of the HSSL at both a system as well as circuit level. As stated earlier, the channel induced degradation is the primary limiting factor during the entire link-design process.



Figure 2.6: 1Gbps Backplane Link Eye Diagram



Figure 2.7: 10Gbps Backplane Link Eye Diagram

2.3.5 Equalization

Equalization is a method of combatting the detrimental effects of intersymbol interference (ISI) caused by the bandlimited channel. Equalizers are typically implemented as linear or non-linear adaptive filters. Equalization performed before the channel is referred to as pre-emphasis and basically involves passing the TX signal through a filter whose transfer function is the inverse of the channel transfer function. Conversely, equalization at the RX end is used to undo the distortion incurred in the received signal due to the channel loss and dispersion. Most RX equalization schemes are adaptive and

are implemented using DSP techniques to cancel out the channel loss from the received data-bits.

2.3.6 Clock and Data Recovery (CDR)

A CDR as the name suggests is responsible for extracting the clock information of the transmitter from the received signal. In modern HSSLs, a clock-recovery mechanism is essential at the receiver end because TX clock information is typically embedded inside the incoming data pulse-stream at the receiver input. At heart, a CDR is essentially a modified PLL circuit wherein the phase-detector now has to sample the incoming data-stream and extract both data and phase information from it [3]. The PD of the CDR detects the transitions in the received data-stream and the VCO generates a periodic clock that drives the decision circuit within the PD to retime the distorted received data and then regenerates the system clock with lower skew and jitter. CDR design is a lot more intricate than PLL design because in the case of the CDR, loop bandwidth is often very small and governed by the jitter tolerance specifications of the system, meaning there is not much room for VCO phase-noise reduction. The most common implementation of a CDR (as shown in Figure 2.8) includes a regular PLL loop to track the exactfrequency of the TX clock, a phase-tracking loop with special phase-detector to produce the retimed data and a common VCO to output a low-jitter, phase-noise replica of the TX clock.



Figure 2.8: Typical Clock and Data Recovery Unit Implementation

2.3.7 Deserializer

The deserializer circuit, as the name suggests, converts the input serial bitstream data back into its original parallel bus form. It is also a completely digital block and it succeeds the RX driver circuit. Basically, the deserializer is just a demultiplexer circuit that is driven by the clock that is recovered by the CDR.

2.3.8 Coding Schemes

Nonreturn-to-zero (NRZ) pulses are commonly used as the basis function for discrete data transmission. The response of the channel to the NRZ pulse is defined as the pulse response and is traditionally used to analyze and model the effects of a channel on data transmission and also in the design of equalizers in the case of channels with large attenuation at the frequency of interest. Apart from NRZ signaling, designers can also implement advanced modulation techniques for faster, robust signaling. For example, multilevel PAM like PAM-4 has much higher spectral efficiencies and can transmit two bits per symbol. This enables the transmission of an equivalent amount of data in half the channel bandwidth. In modern serial links along with the signaling schemes, some amount of encoding is also present in the datastream. Most commonly used encoding schemes are 8B/10B and 16B/20Bwherein 10bits or 20bits are sent but only 8bits/16bits are actual meaningful data bits, and this is powerful as it improves the BER. The only drawback of encoding is that it further adds complexity to the transceiver design as a encoder/decoder circuit needs to be designed and more bits need to be sent through the same bandwidth-limited channel.

2.3.9 HSSL Figures of Merit

HSSL performance is limited by the channel as well as the process technology used during circuit design. Since data-rates are scaling faster than the available channel bandwidths, the major constraint in realizing robust, high-speed interfaces is improving the maximum available clock frequency for on-chip synthesis. The channel bandwidth is the major constraint in overall system performance; thus, dealing with channel loss and designing clever equalization techniques are the biggest design challenges in HSSLs today. Robustness therefore is the most important metric of performance for link designers. The primary figures-of-merit (FOM) for HSSLs are bit-error-rate (BER), jitter, crosstalk analysis and timing/noise analysis [1].

BER in modern HSSLs is typically between 10^{-12} and 10^{-15} and it is the main metric used to signify the integrity of the received data-bits. A BER of 10^{-12} means that 1 bit will incur an error along the link when we transmit a total of 10^{12} bits. Measurement/Estimation of BER is one of the fundamental challenges faced by link designers because in order to accurately conclude that the link actually has a BER of the order 10^{-12} , one needs to simulate a random sequence of at least 10^{12} bits which even in current state-of-the art simulators is next to impossible. Therefore, most simulators use statistical means to collectively analyze the effects of deterministic noise sources such as Intersymbol Interference (ISI), supply-noise, timing-jitter as well as random noise sources like white-thermal noise and random jitter when estimating the system BER.

A common method to measure timing jitter is to use eye-diagrams. Eye diagrams are constructed by slicing the time-domain signal waveform into small sections and overlaying them on top of each other such that the resulting shape resembles an 'eye'. The horizontal axis of the eye diagram represents time and is typically one or two symbols wide, and the vertical axis represents the amplitude of the signal. Ideally, we want the eye to be as "open" as possible, since a larger eye opening signifies that there is a large enough margin to meet any voltage and timing requirements needed by the system. Quantitatively speaking, the minimum height and width of the data at the receiver are key metrics for evaluating link performance. As link designers, we want the receiver eye to be wide enough to provide adequate time to satisfy the setup and hold requirement of the flip-flops used, and have sufficient height to ensure that the voltage levels meet v_{il} and v_{ih} requirements of the system in the presence of multiple noise sources. Figure 2.9 [6] shows an example of what sampling an eye with and without jitter means.



Figure 2.9: Eye Diagram Terminology and FOMs

The most prominent source of signal degradation in HSSLs over a bandlimited channel is Intersymbol Interference (ISI). ISI results when a sequence of signals are passed through a channel whose bandwidth is insufficient to allow passage of all the spectral components of the signal. It is a form of signal distortion caused due to reflections, channel resonances, and channel loss/dispersion. Simplest way to understand ISI is to view it as interference between symbols wherein current bit/symbol causes distortion in subsequent/preceding bit/symbol. ISI degrades as data-rates increase and channel bandwidth remains the same, and the only way to combat it is through clever equalization techniques on either the TX, RX ends or both. Another form of interference which is slowly becoming a major hindrance for link designers as data-rates scale is crosstalk (XT), which is a phenomenon occurring due to presence of capacitive as well as inductive coupling between neighboring signal lines in a transceiver. Typically, most of the XT effects are felt at the connector/package levels of a channel where the signal spacings are small compared to the distance between shields [6]. Near-End XT (NEXT) and Far-End XT (FEXT) are the two classes of XT wherein NEXT is defined as the XT due to energy dissipated from coupling between transmitted signal and the reflected signal on the same chip, while FEXT is defined as the XT due to energy dissipated from coupling between transmitted signals of two different chips. NEXT is by far the most deleterious type of XT and the most commonly observed kind because the TX energy levels are typically very high compared to the RX signal levels so the received signal can really be submerged inside it if proper care is not taken.

Finally, the last major metric in calculating the timing margin of a HSSL is the jitter. Characterization of deterministic as well as random timing jitter in a clock output is very important to a link designer. Essentially, jitter is the time-domain variation in the clock-signal as shown in Figure 2.10 [10]. A commonly used method for jitter calculation is to close either side of the eye horizontally by the amount of peak clock jitter. While this method can be helpful in evaluating the effects of jitter at the receiver end, we will show in this paper that this is an overly optimistic approximation of noise margin degradation for transmitter jitter. Due to the need for integration of clock generators such as PLLs in large digital chips, clock jitter is dominated by power-supply and substrate noise, both of which do not scale with technology. Therefore, as data rates increase, bit-periods become shorter and the performance of multi-gigabit links will be limited by the clock jitter, thereby initiating the importance of accurately analyzing the effects of clock jitter on high-speed serial links. Figure 2.11 [5] provides a summary of common jitter profiles in a typical serial link.



Figure 2.10: Timing Jitter Example



Figure 2.11: Summary of Common Jitter Profiles

CHAPTER 3

PLL THEORY AND BACKGROUND

3.1 PLL Applications

Phase-Locked Loops (PLLs) are one of the most fundamental and ubiquitous circuits found in any communications (wireless, wireline) and high-speed digital systems. Monolithic CMOS implementation of PLLs has gained lots of popularity over the last few decades due to an insatiable demand for high performance digital systems. Most common uses of a PLL are in the form of frequency synthesizers and carrier/clock recovery circuits both in the RF domain as well as the high-speed digital domain.

3.2 Basic PLL Building Blocks

3.2.1 Phase/Phase-Frequency Detector (PD or PFD)

In a PLL, unlike many other feedback systems, the variable of interest changes dimension around the loop: it is converted from phase to voltage (or current) by the phase detector, processed by the LPF as such, and converted back to phase by the VCO. In the lock condition, the input and output frequencies are exactly equal, regardless of the magnitude of the loop gain (although the phase error may not be zero). This is an extremely important property because many applications are intolerant or even small (systematic) differences between the input and output frequencies [15].

The PD compares the phase of the output signal with the phase of the reference signal and develops an output signal that is approximately proportional to the phase-error Φ_e . The output voltage of the PD is proportional to the phase-difference between the reference signal and the output signal. The

PD serves as an "error'-amplifier" in the feedback loop, thereby minimizing the phase-difference, $\Delta \phi$, between the reference signal, $V_{ref}(t)$ and the oscillator output signal, V_{out} . The loop is considered to be "locked" if $\Delta \phi$ is constant with time, a result of which is that the input and output frequencies are equal. In locked condition, all the signals in the loop have reached steady state and the PLL operates as follows. The phase detector produces an output whose DC value is proportional to $\Delta \phi$. The low-pass filter suppresses high-frequency components in the PD output, allowing the DC value to control the VCO frequency. The VCO then oscillates at a frequency equal to the input frequency and with a phase-difference equal to $\Delta \phi$. Thus, the LPF filter generates the proper control voltage for the VCO. The VCO phase can be seen to be an initial condition of the system, as it is independent of the initial conditions in the LPF. Whenever two frequencies become equal at a point in time and $\Delta \phi$ has not established the required control voltage for the VCO, the loop will continue the transient, temporarily making the frequencies unequal again. In other words, both "frequency-acquisition" and "phase-acquisition" must be completed. This behavior is to be expected because for lock to occur again, all the initial conditions of the system, including the VCO output phase, must be updated [15].

The biggest pitfall of using just a PD is that it does not capture any step changes in frequency; thus, in order to be able to track both phase and frequency we need to use a phase-frequency detector (PFD). The purpose of a PFD is to compare the reference clock signal and the VCO output clock after division in both phase and frequency. These frequencies are generally denoted by F_{REF} and F_{VCO} respectively. The basic structure can be divided into logic control part and a charge pump. The charge pump is a current source in series with a current sink and the output node is like a switch that resides in between the source and sink. The logic part consists of two D-Flip-Flops (DFFs) and the outputs of these DFFs control the switch of the charge pump. Conceptually the PFD can be viewed as a state machine with three states. The initial state is 0 and both DFFs will be reset if VCO and reference signal are both high. In state -1 only current sink is turned on and sinks charge out of the load, thereby decreasing the output voltage; in state 0 current source and current sink are turned off so no charge is injected or extracted out of the output node, thereby keeping the output voltage unchanged. In state 1 only current source is turned on so charge can

be injected into the output node, thereby increasing the output voltage. The state transitions are controlled by the edges of VCOs output and reference signal; thus it is clear that the PFD is a purely digital circuit.



Figure 3.1: PFD Operation



Figure 3.2: PFD Functionality

Figures 3.1 and 3.2 show the block diagram of a PFD and demonstrate its functionality. Essentially, when the reference clock is faster than the divider clock, UP signal is High, DN signal is Low and vice versa. Note that when both the reference and divider clocks are synchronized both UP and DN signals are set to be High. The phase frequency detector (PFD) is a circuit that linearly translates the phase difference into voltage signals. The ideal average input/output relationship should be:

$$V_e = K_{PD} \times \phi_e \tag{3.1}$$

where
$$|\phi_e| < 2\pi$$

 K_{PFD} is defined as the PFD gain.

3.2.2 Charge-Pump (CP)

The charge pump is the device that translates the digital voltage signals generated from PFD into a current signal. Since the voltage controlled oscillator needs a stable voltage to control the oscillating frequency, a charge storage capacitor is needed. In order dump enough charge into the capacitor, a charge pump is needed here. Together with the PFD the s-domain transform becomes the following:

$$K_{PFD} = \frac{i_{Charge Pump}}{2\pi} \tag{3.2}$$

3.2.3 Loop-Filter (LF)

PLLs act as high-pass filters so the purpose of the loop filter is to filter out the high-frequency components from the output of the PFD. Typically, loopfilters are just simple passive RC networks whose main objective is to filter out the high-frequency noise data from the PFD output.

3.2.4 Voltage Controlled Oscillator (VCO)

VCOs are the most important and complex component of the overall PLL design. The essential idea behind a VCO design is to generate a clock signal based on the Barkhausen criteria for oscillation which states that the magnitude of the VCO transfer function at the oscillation-frequency is 1 while the phase is -180 degrees. Two most popular VCO topologies whose sample architectures are ring-based and LC-tank based. Due to the superior noise performance we chose to design a LC-Tank based VCO. VCO is the device that generates the target clock. Ideally, its output frequency should be linearly related to the input control voltage. The Laplace transform function of the VCO is derived as follows:

$$\omega_{out}(t) = K_{VCO} v_{ctrl}(t) \tag{3.3}$$

$$\mathcal{L}[\omega_{out}(t)] = \omega_{out}(s) = K_{VCO}v_{ctrl}(s)$$
(3.4)

$$\phi_{out}(t) = \int_{0}^{t} \omega_{out}(\tau) d\tau = \int_{0}^{t} K_{VCO} v_{ctrl}(\tau) d\tau$$
(3.5)

$$\mathcal{L}[\phi_{out}(t)] = \phi_{out}(s) = \frac{\omega_{out}(s)}{s} = \frac{K_{VCO}v_{ctrl}(s)}{s}$$
(3.6)

Thus, the Laplace transform function for the VCO is:

$$H_{VCO}(s) = \frac{\phi_{out}(s)}{v_{ctrl}(s)} = \frac{K_{VCO}}{s}$$
(3.7)

The K_{VCO} is defined as the VCO gain.

3.2.5 Divider

A frequency divider is needed to produce a clock signal that runs many times faster than the reference clock. The PFD input clock and reference clock have to be synchronized for PLL to be in locked condition. In order to perform this task we use a fractional-N divider circuit, which divides the VCO clock by the highest power of 2 factor to synchronize reference clock signal and the divider output clock.

3.2.6 Analysis of a PLL in Locked-State

The open-loop transfer function of the PLL is equal to $H_O = K_{PD}G_{LPF}(s)\frac{K_{VCO}}{s}$, yielding the closed-loop transfer function $H(s) = \frac{\Phi_{out}(s)}{\Phi_{in}(s)} = \frac{K_{PD}K_{VCO}G_{LPF}(s)}{s+K_{PD}K_{VCO}G_{LPF}(s)}$. In its simplest form, a low pass filter is implemented to have the transfer function $G_{LPF}(s) = \frac{1}{1+\frac{s}{\omega_{LPF}}}$, where $\omega_{LPF} = \frac{1}{RC}$. Thus, for a PLL containing a first-order LPF the closed-loop response is represented as $H(s) = \frac{K_{PD}K_{VCO}}{\frac{s}{\omega_{LPF}}+s+K_{PD}K_{VCO}}$ indicating that the system is of second-order, where one pole is contributed by the VCO and the other by the LPF. Here, $K = K_{PD}K_{VCO}$ is called the loop-gain and expressed in rad/s. In order to understand the dynamic behavior of the PLL, the denominator of the second-order closed-loop response is converted to a form commonly used in control theory: $s^2 + 2\zeta\omega_n s + \omega_n^2$, where ζ is the damping factor and ω_n is the natural frequency of the system. Therefore, the closed-loop response can now be expressed as $H(s) = \frac{\omega_n^2}{s^2+2\zeta\omega_n s+\omega_n^2}$, where $\omega_n = \sqrt{\omega_{LPF}K}$ and $\zeta = \frac{1}{2}\sqrt{\frac{\omega_{LPF}}{K}}$. Note that ω_n is the geometric mean of the -3dB bandwidth of the LPF and the loop-gain, and thus an indicator of the gain-bandwidth product of the loop. The damping factor is inversely proportional to the loop gain. Typically, in a well designed second order system, ζ is usually greater than 0.5 and preferably equal to $\frac{\sqrt{2}}{2}$ so as to provide an optimally flat response. Thus, K and ω_{LPF} cannot be chosen independently; for example if $\zeta = \frac{\sqrt{2}}{2}$, then $K = \frac{\omega_{LPF}}{2}$. If $s \to 0$, we note that $H(s) \to 1$; i.e. a static phase shift at the input is transferred to the output unchanged. We can examine the "phase error transfer function" defined as $H_e(s) = 1 - H(s) = \frac{\Phi_e(s)}{\Phi_{in}(s)} = \frac{s^2 + 2\zeta\omega_n s}{s^2 + 2\zeta\omega_n s + \omega_n^2}$ which drops to 0 as $s \to 0$ [15].

3.2.7 PLL Characteristics and Figures of Merit

The most important metrics of a PLL are Order, Type, Hold-In range, Lockin range, and Pull-in range. The order of a PLL is determined by the number of poles in the loop while the type is determined by the number of integrators. The VCO always has a pre-existing pole because it generates frequency from phase via an integration; thus every PLL is at least of order 1 and type 1. As the loop-filter poles increases and the PLL order and type increases as well and higher the type, the better the PLL is at tracking both frequency and phase. For instance, a type 2 PLL is capable of tracking both a step change in phase as well as frequency with zero steady-state phase error while a type 1 PLL can only track a step change in phase.

Hold-In range of a PLL is a measure of the DC loop gain and the range of frequencies under which the PLL can maintain a lock. Lock-In range is a measure of the range of frequencies under which a PLL can acquire lock without slipping any clock-cycles. Finally, the Pull-In range is the measure of the range of frequencies for which the PLL can acquire lock by missing a few clock cycles. It is important to note that the hold-in range is the largest of the three and lock-in range is the smallest of the three metrics.

CHAPTER 4

PLLs IN CLOCKING CIRCUITS

4.1 Charge-Pump (CP) PLLs Overview



Figure 4.1: Charge-Pump PLL Block Diagram

Figure 4.1 shows the basic building blocks of a CPLL. Charge-Pump PLLs offer many advantages over the classical voltage phase-detector PLL including an infinite pull-in range and zero steady-state phase error. CPLLs also allow one to use a passive filter and still have many of the benefits of using an active filter with the voltage phase detector. The exception to this case is when the VCO tuning voltage needs to be higher than the PLL can supply; in this case, an active filter is necessary [15].

Phase-frequency detectors with charge-pump combination offer several advantages over the voltage charge pump and have all but replaced it. The PFD and CP blocks are universally present in every PLL based synthesizer chip. Using this approach completely bypasses issues of steady-state phase error and hold-in range [10].

4.2 CPLL Linear Model and Analysis



Figure 4.2: Linear s-Domain Model for Charge-Pump PLL

Figure 4.2 shows the linear s-domain model for CPLLs. From the previous section we can now define the open loop transfer function as follows:

$$LG(s) = K_{PD} \cdot F(s) \cdot \frac{K_{VCO}}{s}$$

$$s + \frac{1}{PC}$$
(4.1a)

$$= K_{PD} \cdot K_{VCO} \cdot \frac{s + \frac{s}{RC_1}}{C_2 s^2 \left(s + \frac{C_1 + C_2}{RC_1 C_2}\right)}$$
(4.1b)

From the open loop gain we notice that

$$\omega_z = \frac{1}{RC_1}; \ \omega_{p1} = \omega_{p2} = 0; \\ \omega_{p3} = \frac{C_1 + C_2}{RC_1C_2}$$
(4.2)

The phase margin will be:

$$\phi_M = \arctan\left(\frac{\omega_{ugb}}{\omega_z}\right) - \arctan\left(\frac{\omega_{ugb}}{\omega_{p3}}\right)$$
 (4.3)

where ω_{ugb} is the open loop unity gain bandwidth and $\omega_z < \omega_{ugb}$.

In order to achieve maximum phase margin, the value of C_1 and C_2 have to be chosen carefully. To calculate the expression of ϕ_{M_max} we take the first order derivative of Eq. 4.3 with respect to ω_{ugb} and equate the result to zero, such that:

$$\omega_{ugb} = \omega_z \sqrt{\frac{C_1}{C_2} + 1} \tag{4.4}$$

Subsequently,

$$\phi_{M_max} = \arctan(\sqrt{\frac{C_1}{C_2} + 1}) - \arctan(\frac{1}{\sqrt{\frac{C_1}{C_2} + 1}})$$
(4.5)

The design procedure of the loop filter is as follows:

1. Choose desired bandwidth ω_{ugb} , phase margin ϕ_M and resistor R according to specification. Then calculate the K_c from Eq. 4.6:

$$K_c = \frac{C_1}{C_2} = 2(\tan^2(\phi_M) + \tan(\phi_M \sqrt{\tan^2(\phi_M) + 1}))$$
(4.6)

2. From Eq. 4.4 we have:

$$\omega_z = \frac{\omega_{ubg}}{\sqrt{\frac{C_1}{C_2} + 1}} \tag{4.7}$$

$$C_1 = \frac{1}{\omega_z R}; C_2 = \frac{C_1}{K_c}; \tag{4.8}$$

3. From a forementioned equations, we can determine the value for I_{CP} :

$$I_{CP} = \frac{2\pi C_2}{K_{VCO}} \cdot \omega_{ugb}^2 \cdot \sqrt{\frac{\omega_{p3}^2 + \omega_{ugb}^2}{\omega_z^2 + \omega_{ugb}^2}}$$
(4.9)

It is vital to analytically confirm that the PLL will indeed lock when there is a frequency step applied at the input. Without loss of generality assume there is input frequency step $\omega_{in} = \frac{\Delta \omega}{s}$, then $\Phi_{in}(s) = \frac{\Delta \omega}{s^2}$. First, obtain the closed loop transfer function:

$$H_{PLL}(s) = \frac{LG(s)}{1 + LG(s)}$$
 (4.10)

Lastly, define steady state error transfer function:

$$\frac{\Phi_{error}(s)}{\Phi_{in}(s)} = H_e(s) = 1 - H_{PLL}(s) = \frac{1}{1 + LG(s)}$$
(4.11)

Applying the final value theorem, we get the steady state error to be:

$$\Phi_{ss_error}^{F_{step}} = \lim_{s \to 0} s \cdot H_e(s) \cdot \Phi_{in}(s)$$
(4.12a)

$$=\lim_{s\to 0} s \cdot \frac{1}{1+LG(s)} \cdot \frac{\Delta\omega}{s^2}$$
(4.12b)

$$= \lim_{s \to 0} \frac{[RC_1C_2s^2 + (C_1 + C_2)s]\Delta\omega}{RC_1C_2s^3 + (C_1 + C_2)s^2 + K_{VCO}K_{PD}s + 1}$$
(4.12c)

$$=\frac{0}{1} \tag{4.12d}$$

$$= 0$$
 (4.12e)

Eq. 4.12(a) to 4.12(e) indicate that the PLL we have designed can eliminate any steady state phase error and relock when a frequency step is applied at the input [8].

4.3 CPLL Noise-Analysis

The following equations describe the noise-transfer functions of the CPLL and are used in determining the optimal PLL BW.

$$NTF_{IN}(s) = \frac{\Phi_{OUT}(s)}{\Phi_{IN}(s)} = \frac{N \cdot LG(s)}{1 + LG(s)}$$
(4.13)

$$NTF_{DIV}(s) = NTF_{IN}(s) \tag{4.14}$$

$$NTF_{CP}(s) = \frac{\Phi_{OUT}(s)}{i_{CP}(s)} = \frac{2\pi}{I_{CP}} \cdot NTF_{IN}(s)$$
(4.15)

$$NTF_R(s) = \frac{\Phi_{OUT}(s)}{v_R(s)} = \frac{\frac{K_{VCO}}{s}}{1 + LG(s)}$$
 (4.16)

$$S_{\Phi_{OUT}}^{\Phi_{IN}} = S_{\Phi_{IN}} |NTF_{IN}(s)|^2$$
(4.17)

$$S_{\Phi_{OUT}}^{i_{CP}} = S_{i_{CP}} |NTF_{CP}(s)|^2$$
(4.18)

$$S_{\Phi_{OUT}}^{v_R} = S_{v_R} |NTF_R(s)|^2$$
(4.19)

$$S_{\Phi_{OUT}}^{\Phi_{VCO}} = S_{\Phi_{VCO}} |NTF_{VCO}(s)|^2$$
(4.20)

$$G_{LPF}(s) = \frac{1}{1 + \frac{s}{\omega_{LPF}}} \tag{4.21}$$

$$H(s) = \frac{K_{PD}K_{VCO}}{\frac{s}{\omega_{LPF}} + s + K_{PD}K_{VCO}}$$
(4.22)

$$s^2 + 2\zeta\omega_n s + \omega_n^2 \tag{4.23}$$

$$H(s) = \frac{\omega_n^2}{s^2 + 2\zeta\omega_n s + \omega_n^2} \tag{4.24}$$

$$\omega_n = \sqrt{\omega_{LPF}K}, \zeta = \frac{1}{2}\sqrt{\frac{\omega_{LPF}}{K}}$$
(4.25)

$$\zeta = \frac{\sqrt{2}}{2}, \implies K = \frac{\omega_{LPF}}{2} \tag{4.26}$$



Figure 4.3: Loop-Gain and Phase-Margin Response



Figure 4.4: CPLL Output Noise Model



Figure 4.5: CPLL Output Noise Simulation

Figure 4.3 [8] shows the typical loop-gain and phase-margin plot for a CPLL. Recall that phase-margin is the difference in phase between $-180 \,^{\circ}\text{C}$ and the phase value corresponding to ω_{ugb} . Figure 4.4 shows the typical noise-profile for each component in a CPLL based Integer-N synthesizer [12]. Figure 4.5 shows the noise-transfer function characterization for the PLL using the above equations implemented in MATLAB. The beauty of this analysis is that it accurately predicts what the noise-profile for the PLL will look like so that the designer can determine the BW specifications for the PLL from which the rest of the Loop-Filter and VCO specifications can be determined.

CHAPTER 5

PLL BASED CLOCK GENERATOR

5.1 PFD

Figure 5.1 shows the NAND PFD implementation used in the design of the PLL used in this thesis. In Figure 5.2 the transistor-level implementation for each of the circuits shown in Figure 5.1 are displayed with the appropriate sizing.



Figure 5.1: NAND PFD Implementation


Figure 5.2: PFD Transistor Level Circuit Blocks

One of the major challenges during the design of an efficient PFD circuit is the "dead-zone" problem. "Dead-zone" is refers to the region wherein there is no output for inputs. It is equal to the sum of the on-times of the pullup/pull-down switches in the charge-pump and is typically a problem because the presence of dead-zone causes the PLL to operate in "open-loop" when the phase-error is zero. One method to overcome the dead-zone issue is to ensure that the PFD generates equal UP/DN pulses whose width is larger than the switch-on time of the CP switches. The NAND-PFD implementation is one example of a PFD circuit where the dead-zone issue is avoided. The D-Flip-Flops are designed using cross-coupled NAND-latches and even though this uses up a lot of on-chip area and burns a lot of power in the PFD circuit, the large delay in UP/DN feedback paths allows the pulse-widths to be just larger than the switch-on time of the PMOS/NMOS transistors that act as the pullup/pull-down switches in the CP. The maximum operating frequency of a PFD circuit is determined by the reset path delay such that $F_{max} < \frac{1}{2T_{RST}}$. In the case of a NAND PFD circuit, $T_{RST} = 2T_{NAND2} + T_{NAND4}$, where we intentionally design the NAND4 circuit to have a high delay to minimize the reset period.

5.2 CPs

Figure 5.3 shows the CP implementation used in the design of the PLL used in this thesis. The transistor-level implementation is also displayed with the appropriate sizing.



Figure 5.3: Bootstrapped Charge-Pump Implementation

The charge-pump circuit needs to be carefully designed in because it is the main contributor to low-frequency PLL noise. The current mismatch in the charge-pump leads to static-phase offset and causes ripples in the control voltage, thereby creating a 'jittery' VCO output clock. Thus, it is important to design a CP circuit that has equivalent pull-up and pull down currents and equal on-time for the PMOS/NMOS switches. Though several CP architectures exist, a 'Bootstrapped' CP design is used in the clock-generating PLL studied in this thesis. The advantage of the Bootstrapped architecture is that it allows differential current steering, it can operate with low-swing UP, DN signals. It is thus very prominent in PLLs that use high-speed reference clock signals. The term 'bootstrapped' are appropriate because the voltage following op-amp between the pull-up and pull-down current networks ensures that an equal voltage level is maintained on either ends such that the pull-up current is equal to the pull-down current.

5.3 LF



Figure 5.4: Loop-Filter Implementation

Loop-Filter is designed using the design-procedure described in Chapter 4 in the CPLL design procedure algorithm. The algorithm was implemented in MATLAB to choose the values shown above in Figure 5.4.

5.4 VCOs



Figure 5.5: Single-Ended 3-Stage Ring Oscillator

A 3-Stage ring-oscillator is implemented with a driver inverter (as shown in Figure 5.5 with full transistor-level sizing) whose size is fixed such that the input capacitance seen by the divider remains constant while the VCO frequency is changing. M8 and M9 act as the pull-up resistors, i.e. they are PMOS transistors that are biased to be in triode/resistive region. In order to ensure the oscillation starts, the gate of M9 is driven to ground while gate of M8 is driven by the control voltage which alters the phase-delay between the ring to vary to the oscillation frequency.

5.5 Divider



Figure 5.6: TSPC Based D Flip-Flop Architecture



Figure 5.7: Divider Architecture

The divider circuit consists of 3-DFFs that are connected together in the manner shown in Figure 5.7 to realize a divide-by-8 operation. Division in binary is essentially a left-shift operation; thus, tying the outputs of each DFF to clock input of the next while connecting the input to the inverse of output in a feedback ensures a left-shift operation. Since the PLL output frequency is in the GHz range, the DFF design is very critical. To realize a fast DFF with low clock-skew and delay a TSPC (True-Single Phase Clock) architecture (as shown in Figure 5.6) is employed. The basic idea is that when CLK is high transistors M1 and M9 are ON/OFF respectively and vice versa, thereby preserving the state except when CLK goes from low to high, in which case the output follows the data-input signal denoted by D.

CHAPTER 6

BEHAVIORAL LEVEL SIMULATION

6.1 Why Behavioral Modeling?

Traditionally SPICE is used as a common simulation engine to simulate analog/mixed-signal circuit. However, when simulating large networks the simulation times can become extremely long, thereby limiting the allowed design revisions to the circuit designer. It is very tedious to describe the behavior of a circuit using SPICE unless the complete physical transistorlevel structure of the circuit is known to the designer. Furthermore, the SPICE simulation process is very technology dependent in that with technology scaling the SPICE models need to be updated as the older models become obsolete and invalid for accurate simulation. The aforementioned design process has remained virtually the same over the past few decades and even though the digital design synthesis process has progressed significantly by incorporating electronic system-level (ESL) design automation techniques, the mixed signal design process is very slow, laborious and therefore errorprone. Digital design engineers, though working with millions of transistors, have been able to automate the design flow, but analog designers have been unable to do so even though most analog circuits only consist of tens of thousands of devices.

6.2 Why Verilog-AMS?

Verilog-AMS is a high-level Hardware Description Language (HDL) used to describe the structure and behavior of analog and mixed-signal systems. It is an extension to the IEEE 1364 Verilog HDL standard and is very powerful in providing fast prototyping capabilities for mixed-signal systems. The key advantage of circuit modeling using Verilog-AMS is that it provides a single language and simulator ecosystem that can be shared between analog, digital and system-level designers. Verilog-AMS leverages the superior speed and capacity offered by traditional Verilog and allows event-driven capabilities within analog model simulation, making it an attractive choice when simulating highly complex mixed-signal circuits such as PLLs, CDRs, ADCs, and DACs. The only pitfall of using Verilog-AMS is that it cannot replace traditional transistor level SPICE simulation completely as it does not have synthesis capabilities like its digital counterpart Verilog. However, at the onset of the design phase, using Verilog-AMS for circuit modeling is very powerful for a mixed-signal circuit/system design engineer as it offers fast prototyping/verification for behavioral level simulation, thereby expediting the time-to-market for the system.

Verilog-AMS combines both Verilog-D and Verilog-A including a few additional mixed-signal constructs to provide a HDL language capable of performing truly mixed-signal simulation. Cadence has been the front-runner in promoting the language making it an industry standard, and has led the majority of the advancement efforts ever since its release in 2003. The power of Verilog-AMS simulator in Cadence Virtuoso is that it can perform cosimulation among behavioral analog/digital blocks described by corresponding Verilog-A and Verilog-D models respectively as well as transistor-level circuit blocks by running the Spectre simulation. When a circuit consisting of transistor-level circuit elements, analog behavioral modules written in Verilog-A and digital behavioral modules written in Verilog-D is simulated, the AMS simulator in Cadence partitions the testbench into analog and digital components. The simulator then merges the analog simulation results from Spectre with the digital simulation results from NC-SIM and the resulting output is plotted just like that in the case of traditional Spectre simulation [4].

6.3 Basic Verilog-A/AMS Syntax

A typical skeleton of a Verilog-AMS code is shown in Figure 6.1 where the main components of a Verilog-A/AMS code are listed.

```
1 `include "disciplines.vams"
 2
 3 module name(inputs,outputs)
       parameter real var = 0;
 4
 5
       input in1;
 6
       output out1, out2;
 7
       electrical out1,out2;
 8
9
       analog
10
           begin
11
           ----code logic-----
12
13
14
           end
15 endmodule
```

Figure 6.1: Verilog-AMS Sample Code

In the first line of the sample code shown in Figure 6.1 [4], we include the 'disciplines.vams' header file. This file is a collection of physical signal types that are commonly used in Verilog-AMS and are thus referred to as 'natures'. Electrical disciplines consist of 'voltages' and 'currents' and are used most commonly during mixed-signal system modeling where 'voltage' and 'current' are 'natures'. Every Verilog-AMS component is defined as a 'module' and modules are the basic building blocks of any given Verilog-AMS files as they describe the component being modeled. Ports are the points where connections are made to the given component. Every port is required to have a direction associated with it, and by default in Verilog-AMS language there are three types of ports: input, output and inout. The keyword **electrical** signifies that the signals associated with the ports described as electrical are of 'voltage' and 'current' natures. Additionally, **analog** is the keyword after which point the Verilog-AMS compiler starts actual modeling as the logic/process starts after the 'analog begin'. Finally, every Verilog-AMS component code should end with the word **endmodule** as it signifies the point at which the compiler stops parsing of the code [4].

6.4.1 PFD+CP

- 1. Create a new library and name it 'PLLBehav'. Now within the Library Manager window, click on $File \rightarrow New \rightarrow Cell \ View$ and call the new VerilogAMS file pfd. Choose the 'VerilogAMSText' option from the drop-down Menu as shown in Figure 6.2. Click 'OK' and a text editor window will open up.
- 2. Figure 6.3 shows the 'PFD' code used in the design. The PFD is a completely digital circuit; thus, this code is essentially in Verilog-D syntax where the UP, DN signals are generated by comparing the rising edges of the flip-flops that have a CLK signal of F_{REF} and F_{DIV} respectively.



Figure 6.2: PFD Verilog-AMS Code Setup

```
1 //Verilog-AMS HDL for "BehavPLL", "pfd" "verilogams"
 3
   `include "constants.vams"
 4
    `include "disciplines.vams"
 5 `timescale 10ps / 1ps
 6
 7 module pfd (up,dn,upb, dnb,fref,fdiv);
 8
     input fref;
input fdiv;
 9
10
     output up,upb,dn, dnb;
11
12
13 wire fv_rst, fr_rst;
14 wire reset;
15 reg q0, q1;
16
17 assign fr_rst = reset | (q0 & q1);
18 assign fv_rst = reset | (q0 & q1);
19 assign reset = fref & fdiv;
20
21 always @ (posedge fdiv or posedge fv_rst) begin
22 if (fv_rst) q0 <= 0; else q0 <= 1;
23 end
25 if (fr_rst) q1 <= 0; else q1 <= 1;
26 end
24 always @ (posedge fref or posedge fr_rst) begin
27 assign up = q1;
28 assign dn = q0;
29 assign upb = ~q0;
30 assign dnb = ~q1;
31 endmodule
```

Figure 6.3: PFD Verilog-AMS Code

3. Once you have written the code as shown in Figure 6.4, save and exit the text editor. A pop-up window like Figure 6.4 will open up. Click 'Yes' to generate the symbol for the 'pfd'.



Figure 6.4: PFD Verilog-AMS Symbol

4. Within the 'PLLBehav' library you created above, click on $File \rightarrow New \rightarrow Cell \ View$ and call the new schematic pfd_tb as shown in Figure 6.5. Double-click on this cell-view and a schematic window will open up.

$\Theta \circ \circ$	X New File
File	
Library	PLLBehav 🔽
Cell	pfd_tb
View	schematic
Туре	schematic 🔽
Application	
Open with	Schematics L
🗌 Always use	this application for this type of file
Library path fil	e
/home/Werild	og/ckt180/cds.lib

Figure 6.5: PFD Verilog-AMS Schematic

5. In order to create a circuit in the schematic editor, we need to add 'instances' or circuit-components like transistors, supply nets and wires. To add an instance press I from your keyboard. This will open up a 'Component Browser'. Choose the 'PLLBehav' library and within it select the symbol for 'pfd'. Repeat the same process to add the 'vpulse' components found in the 'analogLib' library. Make sure the 'fref' and 'fdiv' sources have a 100ps delay between each other. Figure 6.6 shows what your test-bench schematic should look like.



Figure 6.6: PFD Verilog-AMS Testbench

6. When simulating Verilog-AMS files in Cadence Virtuoso, we need to create a 'config' file whose job is to link the analog test-bench sources and the verilog simulation engines together. In order to do so, within the 'pfd_test' cell-view click on $File \rightarrow New \rightarrow Cell \ View$ and call the new config pfd_test as shown in Figure 6.7. Double-click on this cell-view and a New Configuration window will open up. Click on 'Use Template', choose the AMS template and configure the setup as shown in Figure 6.8(a). Finally the configuration setup will look like that

shown in Figure 6.8(b), so click on 'Save' and press 'Open'. Now a window like the schematic view will open up but this time it will have config in the title.

00	X New File
- File	
Library	PLLBehav 🔽
Cell	pfd_test
View	config
Туре	config 🔽
Application	
Open with	Hierarchy Editor 🧧
🔲 Always use th	is application for this type of file
Library path file	
/home/Verilog	/ckt180/cds.lib
	OK Cancel Help

Figure 6.7: Config File Creation

00	X New Configuration						
Top Cell							
Library:	PLLBehav 💌						
Cell:	pfd_test	000	X Virtuoso® Hi	erarchy Editor: New Confi	guration (Save I	veeded)	
View:	schematic	File Edit View Pit	igins <u>H</u> elp			cad	ence
- Global Bi	ndings		' 🕸 🥱 🥐 🕕		01-1-1-0		0.5
Library Li	et mulih	Top Cell		(78A)	Global Bindinj	js	00
View Lieb	audamuorilaa cahamatia uarilaan ukat ukatama umaal	Library: PLLBehav			Library List	myLib	<u> </u>
Class List	systemvening schemalic veninga vitu vitualiis wear	Cell: pfd_test			View List:	eriloga vhdl vhdlams wreal	
Stup Lise	specine	View: schematic			Stop List:	spectre	C
Constrain	f List	Onon Edit			Constraint List		
Descripti	DN	Open Cont			Constraint List		
Default AMS ner Note: Plea	config view template for OSS-based and Cellview-based Verilog- tilsters in ADE. Ise remember to replace Top Cell Library, Cell, and View	Table View Cell Bindings	Tree View				
heid	s with the actual names used by your design.	Library RehavPL	Cell	View Found	View To Us	e Inherited View List	.t
		PLLBehav	pfd_test	schematic		spectre spice verilog	V
		analogLib	vpulse	spectre		spectre spice verilog	۷
<u> </u>							
	OK Cancel Use Template Help	II Namespace: CDBA	Filters: OFF				
		25(66) >					
	(\mathbf{a})			(b)			

Figure 6.8: Config File Setup

- 7. We will simulate our circuits using Cadence AMS Simulation engine. AMS is capable of simulating Verilog-AMS as well as Spectre components. Spectre is a variant of HSPICE developed by Cadence and provides greater accuracy, speed and flexibility especially when dealing with mixed signal circuits.
- 8. Make sure you first 'Check and Save' your config file and click on $Launch \rightarrow ADE$ to open up the ADE window.

9. Click on $Setup \rightarrow Simulator$ to make sure the Simulator is set to AMS. Select the output nodes and choose a transient simulation for 100ns as shown in Figure 6.9.

\varTheta 🔿 🔿 🛛 🕅 🕅 🕅	rtuoso® Analog	Design Environment (1) – B	BehavPLL pfd_	test co	onfig
Launch Session Setup	<u>A</u> nalyses <u>V</u> ari	ables <u>O</u> utputs <u>S</u> imulation <u>F</u>	<u>R</u> esults <u>T</u> ools	<u>H</u> elp	cādence
💾 🌮 🦵 27	1 🕹 ≻ é	A 🗹 🖻	_		_
Design Variables Name	Value	Analyses Type Enable 1 tran 🖌 0 100n r	Argur noderate	nents	
		Outputs			? # × 🜔
		Name/Signal/Expr	Value Plot	Save	Save Options
		2 fref			yes 🗖
		3 DN		<u> </u>	yes
		4 UP		V	yes
>		Plot after simulation: Auto	Plottir	ng mode	: Replace
≡mouse L:		M:			R
1(4)	Status: Ready	T=27 C Simulator: ams(S	ipectre)Mode: I	batch	State: ams_stateMar20

Figure 6.9: PFD Verilog-AMS ADE Outp Window

10. In the final output waveform shown in Figure 6.10 it is clear that the PFD is functioning correctly. Notice that the UP,DN pulses are appropriately modulated as 'REF' and 'DIV' signals diverge from one another.



Figure 6.10: PFD Verilog-AMS Simulation Output

11. Within the 'PLLBehav' library follow the steps described earlier to create a model for the CP as shown in Figure 6.11 and save the file as 'cp'.

1	//Verilog-AMS HDL for "BehavPLL", "cp" "verilogams"
3	`include "constants.vams"
4	`include "disciplines.vams"
5	`timescale 10ps / 1ps
6	
7	module cp (pout, nout, up, dn);
8	parameter real cur = 1m; // output current (A)
9	input up, dn;
10	output pout, nout;
11	electrical pout, nout;
12	real out;
13	analog begin
14	@(initial step) out = 0.0;
15	if (dn && !up)
16	out = -cur:
17	else if (!dn && up)
18	out = cur:
19	else out = 0:
20	I(pout nout) <+ -transition(out 0.0 10n 10n)
21	end
22	endmodule
22	

Figure 6.11: CP Verilog-AMS Code

12. Figure 6.12 shows the 'PFD+CP' testbench schematic. Create a new schematic named 'cp_test' as well as a config file following the same procedure as the PFD. When simulating using the ADE AMS simulator follow the procedure similar to that shown in Figure 6.9.

	•	• • •	•	•		•	•	•		•	•	•	•	•	•		•	•	•	•	•	•	•	•	•	•	•	•	•	•	• • •	•	•	•	•	•	•	- -	gn	d	•	•
			fd	iv												- Ľ	2			Đ٨	ļ.													Jt	13	•	J					
					e	-										2		-		VF	2				u				- U					Jt		-	· 1 ·			\supset		
	•	·			fr													-																								
J		Į, Į	ø •1=1 •2•⊠		n(‡),),	1=1 2-и	.8																																		
		ť	r=1	00p		t	-=10	10p																																		
		•																																								
				ė.																																						
• •				_ J																																						

Figure 6.12: PFD+CP Verilog-AMS Testbench

13. One of the powerful advantages of behavioral modeling is that we can easily alter values of design variables to modify the functionality of a block. In the 'config' testbench file if you click on the 'CP' block and press **q**, a window as shown in Figure 6.13 will appear. Enter the appropriate value of charge-pump current as per the design objectives.

Apply To Only cur	rent 🧧 instance 🔽		
Show 🗌 syste	m 🗹 user 🗹 CDF		
Browse	Reset Instance Labels	Display	
Property	Value		Display
Library Name	BehavPLL		off
Cell Name	ср		off
View Name	symbol		off
Instance Name	I10		value
	Add Delete	Modify)
User Property	Master Value	Local Value	Display
modelName	ср		off
partName	ср		off
vendorName			off
verilogFormatPr.	PrintBehaveModel		off
CDF Parameter of view	verilogams 🔽		Display
cur	7211		off

Figure 6.13: CP Verilog-AMS Testbench Variable Setup

14. The purpose of the charge-pump is to convert the digital PWM signal outputs from the PFD into a current. As seen in the code and from the final output waveform shown in Figure 6.14, it is clear that the

'PFD+CP' is functioning correctly. When UP is high the current the pull-up current source is on and when DN is high the pull-down current source is on.



Figure 6.14: PFD+CP Verilog-AMS Simulation Output

6.4.2 LF

We use the analog loop-filter as shown in Figure 5.4.

6.4.3 VCO

1. The VCO is the most critical component of the PLL we try to model using Verilog-AMS because it allows us to behaviorally estimate the jitter specifications. Within the 'PLLBehav' library follow the steps described earlier to create a model for the VCO as shown in Figure 6.15 and save the file as 'vco'. Only the white-noise jitter is considered in this design and it is modeled by a Gaussian white-noise probability distribution function.



Figure 6.15: VCO Verilog-AMS Code

2. Figure 6.16 shows the 'VCO' testbench schematic. Create a new schematic named 'vco_test' as well as a config file following the same procedure as the PFD. When simulating using the ADE AMS simulator follow the procedure similar to that shown in Figure 6.9.

		Vc	trl		-							۰.	•		14					
	T	/0.												n,	. 8		Vo	ut	 ~	
	ŧ) ^{vdk}	o=v	etri					•					٩					_>	
	1																			
	Т																			
	ł																			
	\Leftrightarrow																			

Figure 6.16: VCO Verilog-AMS Testbench

3. Just like in the case of the 'CP' in the 'config' testbench file, if you click on the 'VCO' block and press q, a window as shown in Figure 6.17 will appear. Enter the appropriate value VCO design parameters as per the design objectives.

00	X Edit Object Properties	
Apply To On	ly current 🧧 instance 🔽	
Show 🗌	system ⊻ user ⊻ CDF	
Brow	rse Reset Instance Labels Disp	lay
Property	Value	Display
Library Nan	BehavPLL	off 🔽
Cell Name	VCO	off 🔽
View Name	symbol	off 🔽
Instance Na	me I6	value 🔽
	Add Delete	Modify
User Prope	erty Master Value Loo	cal Value Display
interfaceLa	astCh. 21 01:08:30 2014	off
modelNam	e vco	off 🔽
partName	VCO	off 🔽
vendorNar	ne	off 🔽
verilogFor	matPr?rintBehaveModel	off 🔽
CDF Parameter of vi	ew verilogams 🔽	Display
Vmin	0	off 🔽
Fmin	0.9e+09	off 🔽
Fmax	1.8e+09	off 🔽
Vamp	1.8	off 🔽
vtol	1e-09	off 🔽
min_pts_update	32	off 🔽
tran_time	1e-11	off 🔽
jitter_std_ui	0.02	off 🔽
ttol	5e-16	off 🔽
	1.8	off 🗖

Figure 6.17: VCO Verilog-AMS Testbench Variable Setup

4. The VCO circuit is supposed to generate a periodic square-wave output at the desired frequency of interest (as a function of the control voltage) with a certain jitter level which in our case is chosen to be 2% Unit-Interval (UI) of period. From the final output waveform shown in Figure 6.18 it is clear that the 'VCO' is functioning correctly.



Figure 6.18: VCO Verilog-AMS Simulation Output

6.4.4 Divider

1. Divider is essential when designing a clock-generating circuit as we need to scale down the VCO output clock to the reference frequency level such that the two signals can be compared. Within the 'PLLBehav' library follow the steps described earlier to create a model for the Divider as shown in Figure 6.15 and save the file as 'div'. Figure 6.19 shows the code to implement the divider in Verilog.

```
1 //Verilog-AMS HDL for "BehavPLL", "divider" "verilogams"
 3
    `include "constants.vams"
 4
    `include "disciplines.vams"
    timescale 10ps/ 1ps
 5
 7 module divider(out,clk);
        input clk;
output out;
 8
 9
10
        parameter divide_ratio = 8;
        reg out;
integer i=0;
11
12
13
14
15
16
17
18
19
20
21
22
23
24
25
26
27
28
29
30
31
        always@(posedge clk) begin
             if (i < (divide_ratio/2)-1) begin
    out = 0;</pre>
                  i = i + 1;
             end
             else if (i == (divide_ratio/2)-1) begin
                  out = 1;
                  i = i + 1;
             end
             else if (i < (divide_ratio) -1) begin</pre>
                  out = 1;
                  i = i + 1;
             end
             else if (i == (divide_ratio)-1) begin
                  out = 0;
                  i = 0;
             end
       end
32 endmodule
```

Figure 6.19: Divider Verilog-AMS Code

2. Figure 6.20 shows the 'Divider' testbench schematic. Create a new schematic named 'div_test' as well as a config file following the same procedure as the PFD. When simulating using the ADE AMS simulator follow the procedure similar to that shown in Figure 6.9.

		•																	
												in							
	VCC	⊳_ìr	1		÷.,	÷	÷.,				۰.	712	· 0	liví	de¢	but			
								IVIC	ler							-	2.5	dec	
	T M																		
، سر	Ð,	/1= /2:Ø	1.8 i																
	T																		
		•																	
	\rightarrow																		

Figure 6.20: Divider Verilog-AMS Testbench

3. Just like in the case of the 'CP and VCO', in the 'config' testbench file if you click on the 'Divider' block and press **q**, a window as shown in Figure 6.21 will appear. Enter the appropriate value of divide ratio as per the design objectives.

Apply To Only C	urrent 🔽 instance 🔽	
Show 🔲 sys	tem ⊻ user ⊻ CDF	
Browse	Reset Instance Labels Display	
Property	Value	Display
Library Name	BehavPLL	off 🔽
Cell Name	divider	off 🔽
View Name	symbol	off 🔽
Instance Name	13	value 🔽
	Add Delete Modify	
User Property	Master Value Local Value	Display
interfaceLastCl	n. 21 03:02:14 2014	off 🔽
modelName	divider	off 🔽
partName	divider	off 🔽
vendorName		off 🔽
verilogFormatF	Y?rintBehaveModel	off
CDF Parameter of view	verilogams 🔽	Display
divide ratio	8	off

Figure 6.21: VCO Verilog-AMS Testbench Variable Setup

4. The divider circuit is supposed to generate a periodic square-wave output that is fraction of the VCO output frequency. From the final output waveform shown in Figure 6.22 it is clear that the 'Divider' is functioning correctly in that it divides the VCO output signal by a factor of 8.



Figure 6.22: Divider Verilog-AMS Simulation Output

6.4.5 Complete PLL Analysis with Jitter

1. Create a new schematic within the 'PLLBehav' library and name it 'PLL'. Your schematic should look that shown in Figure 6.23. Now create a config file for this setup and at the end your configuration window should look like Figure 6.24.

												æ,		. 1	DΝ																			712												
			unaf																					-11	Ø.											Zate)(6	
-			WIG1	_	<u> </u>	<u> </u>							-			Ľ	-							15		<u> </u>	-i -		-			ilter			<u> </u>	766	<u> </u>									
T VA									· P	fd -					u۲	•					оp					<u>.</u>															YCO	•				6
∲ ∖*	ĿØ																																													
Ŧ۲	2-1.8	· ·																																												
4 9	=16%	Ρ.					5																																							
<u> </u>																											1																			
1																													-	2.																
L.,																														5																
																														2																
																														ě.																
																														<u>\$</u>																
													1.00	÷.,						13																										
				· L		• •				• •			_	JI Y					<u> </u>	-			divi	der			olk.								• •				=	_	_	_		_	roue	1
																											1																			

Figure 6.23: PLL Verilog-AMS Testbench

) 🗁 🗔 🖟	? «J- 🥱 🦿 🛈) 🖬 🤞 🗟 🥥			
p Cell		? 5 ×	Global Bindings		?
brary: BehavPLL			Library List: m	iyLib	٦ <u>с.</u>
			View List: s	pectre spice verilog ver	1
iew: schematic			Stop List:	nectre	
			otop atter	Peene	-
			_		
Open Edit Table View Cell Bindings	Tree View	View Found	Constraint List:	Inherited View I	iet
Open Edit Table View Cell Bindings Library BehavPLL	Tree View Cell	View Found	Constraint List:	Inherited View L	.ist
Open Edit Table View Cell Bindings Library BehavPLL BehavPLL	Tree View Cell Filter PLL	View Found schematic schematic	Constraint List:	Inherited View L spectre spice veril spectre spice veri	ist og
Open Edit Table View Cell Bindings Library BehavPLL BehavPLL BehavPLL	Tree View Cell Filter PLL cp	View Found schematic verilogams	Constraint List:	Inherited View L spectre spice veril spectre spice veril spectre spice veril	.ist og og
Open Edit Table View Cell Bindings Library BehavPLL BehavPLL BehavPLL BehavPLL	Tree View Cell Filter PLL cp divider	View Found schematic schematic verilogams verilogams	Constraint List	Inherited View L spectre spice veril spectre spice veril spectre spice veri spectre spice veri	.ist og og og
Open Edit Table View Cell Bindings Library BehavPLL BehavPLL BehavPLL BehavPLL BehavPLL	Tree View Cell Filter PLL Cp divider pfd	View Found schematic schematic verilogams verilogams verilogams	Constraint List:	Inherited View L spectre spice veril spectre spice veril spectre spice veril spectre spice veri spectre spice veril	.ist og og og og
Open Edit Table View Cell Bindings BehavPLL BehavPLL BehavPLL BehavPLL BehavPLL BehavPLL	Tree View Filter PLL cp divider pfd vco	View Found schematic schematic verilogams verilogams verilogams verilogams	Constraint List:	Inherited View L spectre spice veril spectre spice veril spectre spice veril spectre spice veril spectre spice veril spectre spice veril	.ist og og og og og
Open Edit Table View Cell Bindings EehavPLL BehavPLL BehavPLL BehavPLL BehavPLL BehavPLL analogLib	Tree View Filter PLL Cp divider pfd vco cap	View Found schematic schematic verilogams verilogams verilogams verilogams spectre	Constraint List:	Inherited View L spectre spice veril spectre spice veril spectre spice veril spectre spice veril spectre spice veril spectre spice veril	.ist og og og og og og
Open Edit Table View Cell Bindings EehavPLL BehavPLL BehavPLL BehavPLL BehavPLL BehavPLL analogLib analogLib	Tree View Filter PLL cp divider pfd vco cap res	View Found schematic schematic verilogams verilogams verilogams verilogams spectre spectre	View To Use	Inherited View L spectre spice veril spectre spice veril spectre spice veril spectre spice veril spectre spice veril spectre spice veril spectre spice veril	.ist og og og og og og og
Open Edit Table View Cell Bindings Library BehavPLL BehavPLL BehavPLL BehavPLL BehavPLL analogLib analogLib analogLib analogLib	Tree View Filter PLL cp divider pfd vco cap res vdc	View Found schematic schematic verilogams verilogams verilogams verilogams spectre spectre spectre	View To Use	Inherited View L spectre spice veril spectre spice veril	.ist og og og og og og og og

Figure 6.24: PLL Verilog-AMS Config Setup

2. Using the steps described earlier in this chapter, configure your ADE window as shown in Figure 6.25 and simulate the circuit.

● O O X V	/irtuoso® Analog	Design Environment (2) -	BehavPLL	PLL conf	ĩg	
Launch S <u>e</u> ssion Set <u>up A</u>	<u>∖</u> nalyses <u>V</u> ariable	s <u>O</u> utputs <u>S</u> imulation <u>R</u> es	ults <u>T</u> ools	<u>H</u> elp	cād	d e n c e
🛚 🚰 🧽 🦵 27	🕹 ⊁ 🙆 I	2 🖻	_		_	
Design Variables Name	Value	Type Enable 1 tran ✓ 0 100 c	i conservativ	Arguments e	?)(
		Outputs	I the local in the		? (= × 📈
		1 vout	Value P	lot Save	ves	
		2 vdiv			yes	
		3 vref			yes	=
		4 Vetrl			yes	
		5 DN		I	yes	
		6 UP		V	yes	
		7 freq		I		7
>		Plot after simulation: Auto	F	Plotting mo	de: Replace	
≡mouse L:		M:				R
2(6)	Status: Read	y T=27 C Simulator: am	s(Spectre)M	/lode: batc	h State: ams_st	ateApr2

Figure 6.25: PLL Verilog-AMS ADE Setup

3. The PLL circuit outputs are shown in Figure 6.26. It is clear that the PLL achieves lock within the first 100ns because in the testbench we provide an initial condition of $V_{ctrl} = 0.9V$ and keep the currents at the loop-filter capacitors at an initial condition of 0A. These initial conditions are provided to ensure that the simulation time is small. From the final output waveforms it is clear that the 'PLL' is indeed functioning correctly.



Figure 6.26: PLL Verilog-AMS Simulation Output

4. To simulate the jitter at the VCO output during lock-condition, select the *vout* waveform, click on *Measurements* → *EyeDiagram* and configure the setup as shown in Figure 6.27. Your final output should look like that shown in Figure 6.27 once you click on 'Plot Eye'. The simulated edge-to-edge jitter is 0.96ps which is extremely good. However, it is important to note that this number is not realistic as we have only accounted for random jitter caused by white-noise and the model is only behavioral so any transistor-level non-idealities are not captured. Nevertheless, behavioral modeling is very powerful in performing rapid prototyping of the PLL circuit elements and performs a system level noise/timing budget for the design before delving straight into transistor level design.



Figure 6.27: PLL Verilog-AMS Jitter

CHAPTER 7

TRANSISTOR LEVEL SIMULATION

7.1 What is SPICE?

Simulation Program with Integrated Circuit Emphasis (SPICE) is a generalpurpose circuit simulation program that was originally developed at the University of California-Berkeley to serve as a numerical circuit solver that is capable of performing DC, Transient, as well as AC analyses for electronic circuits. The simulator in general is capable of performing the aforementioned analyses on circuits containing resistors, capacitors, inductors, independent voltage and current sources, dependent sources, lossless and lossy transmission lines, switches, uniform distributed RC lines, and the five most common semiconductor devices: diodes, BJTs, JFETs, MESFETs, and MOSFETs. Many variants of SPICE have been developed since with the most popular ones being HSPICE and Spectre.

7.2 SPICE vs. Spectre

The Spectre circuit simulator is a variant of SPICE that was developed by Cadence to simulate analog and digital circuits at the differential equation level. Although at a high-level the Spectre and SPICE circuit simulators are quite similar in terms of functionality, Spectre directly is not dependent on SPICE and the two simulators also have differing syntax. The parent algorithms for both are primarily the same in that both use the Modified Nodal Analysis (MNA) method involving implicit integration methods, Newton-Raphson, and direct matrix solution, but the source codes are not borrowed from original open-source SPICE. Spectre is optimized for faster speed as well accuracy compared to SPICE and is thus much more reliable and accurate.

7.3 Transient, PSS and PNoise Simulation Overview

Transient response is the time-domain simulation response for a given circuit and is used to study the time-domain behavior of voltages and currents at any given node in a network. It is a powerful analysis method to study amplifier circuits; however, in the case of oscillators it falls short in being able to accurately characterize the harmonic behavior of the outputs. Thus, to study oscillator, mixer circuits or for that matter any circuit that has a time-varying or periodic nature, the Periodic Steady State (PSS) analysis is the preferred method of simulation.

PSS is a large-signal analysis tool and is powerful in accurately determining the approximate small-signal period of the circuit being analyzed. It uses the Iterative Shooting Newton method to algorithmically determine the fundamental frequency of the circuit/system based on the input-source frequency excitation. In PSS, a circuit is evaluated for one period of the target frequency and this period is dynamically adjusted until all node voltages and branch currents fall within a specified tolerance level. Thus, when simulating large networks the PSS simulation often fails to converge and the time-step needs to be manually adjusted. It is also possible that the simulator is just not robust enough for PSS to converge if the time-step is made too small. The first step in a PSS simulation is to perform a transient simulation on the network from time t = 0 to $t = \frac{1}{f_{fund}}$. The next step is then to adjust the time-step adaptively such that the voltage and currents at stabilize within the threshold levels set for the start and stop times of the shooting interval. Figure 7.1 further describes this phenomenon graphically. <u>Note:</u> It is critical to remember that PSS simulation is only valid, and thus will only work, if the circuit/system being analyzed is periodic as the fundamental assumption of PSS analysis is periodicity.



Figure 7.1: PSS Simulation Algorithm

As discussed in earlier chapters, when studying oscillators and PLL circuits the phase-noise is a very important parameter to calculate/simulate. Phase noise is the most significant source of noise in oscillators, and since it is spectrally centered around the fundamental oscillation frequency, methods like filtering cannot eliminate it. PNoise analysis engine within Spectre is equipped to predict the phase-noise, as well as the total-noise profile which includes thermal, flicker and shot noise. Once the PSS simulation for the circuit being analyzed has been completed, the PNoise analysis can be started and it computes the frequency convention, noise-folding and aliasing effects for the circuit/system.

7.4 PLL Simulation in Spectre Using Cadence Virtuoso

This section presents a detailed step-by-step tutorial on conducting transistor level simulations using Cadence Virtuoso's Spectre circuit simulator engine. First, an inverter circuit used inside the PFD is described to illustrate the basic steps required in creating a new schematic and testbench. Second, an in-depth VCO simulation guide illustrates the steps involved in performing PSS and PNoise simulations. Lastly, the full PLL consisting of the PFD, CP, Filter, VCO and Divider blocks is simulated at the transistor level along with the various steps involved in validating lock condition, noise profile and transient response.

7.4.1 Creating a New Schematic

- 1. Create a new library and name it PLL. Now within the Library Manager window click on $File \rightarrow New \rightarrow Cell \ View$ and call the new schematic *inv*. Double-click on this cell-view and a schematic window will open up.
- 2. In order to create a circuit in the schematic editor we need to add 'instances' or circuit-components like transistors, supply nets and wires. Since we use an inverter as an example, recall that we need one NMOS and one PMOS transistor; thus, to add an instance press I from your keyboard. This will open up a 'Component Browser'. Choose the

'analogLib' library. You will notice all the components housed within the 'tsmc18rf' library listed. The key trick to know is that you can search for a specific component from the 'Filter'. Search for 'nmos2v' and follow the steps outline in Figure 7.2.







(b)

Figure 7.2: Inserting NMOS Transistor on Schematic

3. Similarly, following the same steps as (2), add a PMOS transistor to your schematic by choosing the 'pmos2v' transistor from the 'tsmc18rf' library. Your schematic should now look like Figure 7.3.



Figure 7.3: PMOS Transistor

4. In order to add wires to your schematic, press W from your keyboard and make appropriate connections across all transistor elements. Figure 7.4 demonstrates the steps involved in labeling wires with a circuit schematic. This will come in very handy during simulation, especially when dealing with circuits with several components.



Figure 7.4: Inserting Wire Names on Circuit



000	X Virtu	oso® Sch	ematic	Edito	or L E	diting	: TestLi	b inv	schei	matic				
Launch <u>F</u> ile <u>E</u> dit <u>V</u> ie	w <u>C</u> reate	Chec <u>k</u>	O <u>p</u> tions	Mi	grate	<u>W</u> ind	ow <u>H</u> elp	р					cād	епсе
	÷ 0		(Ţ	1	\$	¢ 4	રે ∙	T	T	Q	Q	»	<mark>₽</mark> 5 »
	Worksp	ace: Basic	:			- 2			⊐ 12	16	-R	$\mathbf{T}_{\mathcal{J}}$		Q.
Navigator ?	ē×						, ,							
Y Default							DA							
Name														
M0 (nmos2v)					F	omo	s2v		И1					
OO X Add	Pin						←<	_ ['pc v+2i	n∵ u				
Pin Names IN				· Г				Ľ	, <u> </u>					
Direction input		Bus Exp							:18 Ind	Øn				
Usage schem	atic 🔽	Placeme						י ר	n:1	CI 3.				
Signal Type Signal				\mathbf{i}										
		Vaa		-/	•		~ <u>}</u>							
Aπach Net Expression:		Yes				imc	SZV	ļ,	'nc	h''				
Property Name								\ \	v:2	u.				
Default Net Name				. L				.]	·18	Øn				
Font Height 0.06	25	Font Sty						<u>}</u>	ling	ers	:1			
AL Rotate	deways	Upside						Iľ	m:1					
							Ă	• -'						
Hide Cancel	Defaults	Help												
							. D							
⊪mouse L: mouseAddPt(t)	- 11			M: I	Rotate	90					R: \$	chHiN	/louseF	opUp()
1(3) Point at location poi	nt for the pi	in.										Cm	d: Pin	Sel: 0 //

(b) 000 X Add Pin VDDA Pin Names inputOutput 📕 Bus Expansion 💿 off 🔾 on Direction schematic Placement Usage 🖲 single 🔾 multiple Signal Type signal -Attach Net Expression: 🖲 No 🔾 Yes Property Name Default Net Name stick -Font Height 0.0625 Font Style 🔼 Rotate / Sideways 🛛 🚭 Upside Down 🛛 Show Sensitivity >> Hide Cancel Defaults Help (c)

Figure 7.5: Creating Pin Names

- 5. It is often advisable to add 'Pin' names to each of the IO terminals in a circuit. Thus, to add pins to your schematic press **P** from your keyboard or click on the pin symbol as shown in Figure 7.5 and make appropriate connections across all IO ports. Figure 7.5 demonstrates the steps involved in labeling wires with a circuit schematic. <u>Note:</u> The 'VDDA' and 'GNDA' pins should be chosen to be 'InputOutput' when selecting the 'Direction' during pin creation.
- 6. Finally your schematic should look like Figure 7.6. Now click on 'Check and Save' icon (as shown in Figure 7.7) in the toolbar so that you can move onto the next step of creating a symbol for the inverter schematic.



Figure 7.6: Inverter Schematic



Figure 7.7: Check and Save

7.4.2 Creating a Symbol

1. When dealing with large circuits its often advisable to generate symbols for each sub-circuit in the design and perform all simulations by placing the corresponding symbols in a testbench. Figure 7.8 summarizes the steps involved in generating a symbol from the inverter schematic designed in the previous section.

000	📉 Virtuoso® Schematic Editor L Editing: TestLib inv schematic	
Launch <u>F</u> ile <u>E</u> dit <u>V</u> iew	<u>Create</u> Check Options <u>Migrate Window</u> Help	cādence
	🍱 Instance I 🕜 🖉 🕶 🎢 🖬 🕺 🕄 🕄 🔹	abc 🔎 🖻
Navigator	1. Wire (wide) W 1. Wire (wide) Shift+W	•
▼ Default Q. Search	™ Wire Name L Wirg Stubs and Names Space Net Egression	
Name ← inv ← @ M0 (nmos2v)	Eln P State of the state of	
- 1 GNDA - 1 IN - 1 OUT	Mapping Schematic 952v T M1 Geltview Prom Celtview VODA "Coch" Solder Dot From Celtview VODA "Coch"	
	Note VODA VODA Patchcord Probe OUT fingers:1	
VDDA:P_2	MylliSheet	
Property Editor		
	CNDAT fingers:1	
	e e e e e e e e e e e e e e e e e e e	
mouse L: schSingleSelectPl	10 M: schZoonFit(1.0.0.3) R: schH	iMousePopUpA
2(4) >	v (,)	Cmd: Sel: 0



000	X Cellview From Cellview
Library Name	TestLib
Cell Name	inv
From View Name To View Name	schematic 💌
Tool / Data Type	schematicSymbol 🔻
Display Cellview Edit Options	⊻ ⊻
	OK Cancel Defaults Apply Help
	(b)

Figure 7.8: Generating Symbol from Schematic

2. Once you create the symbol it will pop up in a new window. By default Cadence will generate a rectangular symbol, but you can edit the generated symbol as per your needs. In our case we will edit the symbol shape to make it resemble the traditional inverter symbol used in conventional system design (as shown in Figure 7.9). To edit the shape use the 'Edit Pallete' as shown in Figure 7.9(a) via a red highlighted box.



Figure 7.9: Designing Schematic Symbol

7.4.3 PFD, CP, Filter, VCO and Divider Schematics

1. Using the steps mentioned in the subsections above, create a new schematic as well as symbols for 'nand2', 'nand3' and 'nand4' circuits respectively. Make sure you change the symbol shapes for the *nand* circuits to the conventional symbols shapes.

- 2. Create a new schematic and save it as 'pfd'. Place the 'nand2', 'nand3', 'nand4' and 'inv' symbols in the schematic and connect the four components in the NAND-PFD form as shown in Figure pfd.
- 3. Create a new schematic and save it as 'bias_amp'. Now recreate the biasing op-amp that is part of the charge-pump schematic shown in Figure 5.3.
- 4. Create new schematic and save it as 'cp'. Place the biasing amplifier created in the previous step and recreate the charge-pump circuit shown in Figure 5.3 from section 5.2.
- 5. Create a new schematic and save it as 'filter'. Recreate the loop-filter circuit shown in Figure 5.4 from section 5.3.
- 6. Create a new schematic and save it as 'vco'. Recreate the VCO circuit shown in Figure 5.5 from section 5.4.
- Create a new schematic and save it as 'dff'. Now recreate the biasing D Flip-Flop that is part of the divider schematic shown in Figure 5.6. from section 5.5.
- 8. Create a new schematic and save it as 'div'. Recreate the Divider circuit shown in Figure 5.7 from section 5.5.

7.4.4 Creating a Testbench

Create a new schematic following the steps outlined in the earlier sections and name it 'Tb_pfd_cp_filter'. This will be the testbench schematic from which we will run all our simulations to test that the PFD, CP and the Filter are functioning as expected. Insert 'vdc', 'gnd' and two 'vpulse' from the Component Library by navigating to the 'Analog Parts' library. Figure 7.10 shows the initial conditions to be set for the voltage sources 'vdc', 'vpulse for Vref' and 'vpulse for Vdiv', respectively and Figure 7.11 shows what your testbench schematic should look like at the end of this step.

				00	X Edit Obje	ct Properties	
				Apply To Only cu	irrent 🔽 instanc	e 🔽	
				Show 🗌 syst	tem ⊻ user ⊻ C	DF	
				Browse	Reset Instand	ce Labels Display	Disels
00	X Edit Object Pro	operties		Library Name	analooLib	value	off
Apply To Only curr	rent 🔽 instance 🔽	2	Ê	Cell Name	voulse		off
Show 📃 system	m ⊻ user ⊻ CDF			View Name	symbol		off
Browse	Reset Instance La	bels Display		Instance Name	V1		off
Property	V	alue	Display		bhé	Delete Modify	
Library Name	analogLib		off 🔽	User Property	Master Val	Je Local Value	Displa
Cell Name	vdc]	off 🔽	lvslgnore	TRUE		off
View Name	symbol		off 🔽	CDE Parameter		Value	Dienis
Instance Name	¥2		off	Frequency name for 1/pe	riod	- auto	off
	Add De	lete Modify		Noise file name			off
User Property	Master Value	Local Value	Display	Number of noise/freq pairs	s O		off
Traignore	TRUE			DC voltage			off
CDF Parameter	V	alue	Display	AC magnitude			off
Noise file name	0		off	AC phase			off
vumper of noise/freq pairs	0		off	XF magnitude			off
AC magnitude	T.8 A		off	PAC magnitude			off
AC magnitude			off	PAC phase			off
XE magnitude			off	Voltage 1	0 4		off
PAC magnitude			off	Voltage 2	1.8 ₩		off
PAC phase			off 🔽	Period	4n s		off
Temperature coefficient 1			off 🔽	Delay time			off
Temperature coefficient 2			off 🔽 🚽	Rise time			off
OK Canc	(a)	ply To only c	Lext Help Edit Obje	act Properties	(b)	
OK Canc	(a) (a)	pply To only c or sys Property	Edit Obje uurrent ♀ instan stem ¥ user ¥ Reset Instar	act Properties CDF CDF Value	(b Display)	
OK Cano	(a)	pply To only c orbit or only c orbit or only c sys Browse Property Library Name	Edit Obje Edit Obje uurrent Instan stem Vuser V Reset Instan analogLib	ect Properties ice CDF ice Labels Display Value	(b Display)	
OK Cano	(a)	pply To only c only c orw sys Property Library Name Cell Name	Edit Obje uurrent ♥ Instan stem ♥ user ♥ Reset Instan analogLib vpulse	ect Properties Ice CDF Ice Labels Display Value	(b)	
OK Cano	(a)	pply To only c only c orw sys Property Library Name Cell Name View Name	Edit Obje uurrent 🔮 instan stem 🗹 user 🗹 Reset instan symbol symbol	Act Properties	(b)	
OK Cano	(a)	pply To only c only c orw sys Property Library Name Cell Name View Name Instance Name	Edit Obje surrent instant stem user Reset Instant analogLibly vpulse symbol v2	Act Properties	Display of Control of)	
OK Cano	(a)	pply To only c orw sys Property Library Name Cell Name View Name Instance Name	Edit Obje uurrent ♥ Instan stem ♥ user ♥ Reset Instan vpulse symbol. v2 Add	Act Properties	(b Display of 0 of 0 of 0)	
OK Cano	(a)	Utts Previous N pply To only c tow sys Property Library Name Cell Name View Name Instance Name User Property Ivsignore	Edit Obje C Edit Obje turrent C Instan stem V user V Reset Instar vpulse symbol vz Add Master Va TRUE	Act Properties	Display off P Display off P Display off P)	
OK Cano	(a)	ults Previous N pply To only c tow sys Property Library Name Cell Name View Name Instance Name User Property Ivsignore CDF Paramete	Edit Obje Uurrent Constant stem Vuser Constant Reset Instant Preset Instant Symbol V2 Add Master Va TRUE ar	cct Properties cce cDF cce Labels Display Value Delete Modify due Local Value Value Value Value Value	Display of D of D of D of D of D of D of D of D)	
OK Cano	(a) (a) Fri	Instance Name User Property Library Name Cell Name View Name Instance Name User Property Listgnore CDF Paramete equency name for 1/p	Edit Obje urrent Constant stem Vuser V Reset Instart vpulse symbol V2 Add Master Va TRUE er erlod	cct Properties cce E CDF CDF Value Delete Modify Idue Value	Display off C Display off C Display off C Display off C)	
OK Cano	(a) (a) Fri No	Instance Name User Property Library Name Cell Name View Name Instance Name User Property Listgnore CDF Paramete equency name for 1/pr pise file name	Edit Obje urrent Constant stem Vuser V Reset Instart analogLibl vpulse symbol V2 Add Master Va TRUE eriod	cct Properties cce E CDF CDF Value Delete Labels Display Value Value	Display off C Display off C Off C Display off C Display off C Display off C)	
OK Cano	(a) (a) Fri No Nu	Utts Previous N pily To only c tow system Property Library Name Cell Name View Name Instance Name User Property Ivsignore CDF Paramete equency name for 1/project file name imber of noise/freq pai	Edit Obje User Edit Obje User Instan stem User Reset Instar Reset Instar analogLibl Vpulse synbol V2 Add Master Va TRUE er er o fr 0	Act Properties	Display off D off D off D off D off D Display off D Display off D off D off D)	
OK Cano	(a) (a) Prive the second secon	Utts Previous N pily To only c tow sys Property Library Name Cell Name View Name Instance Name User Property Ivsignore CDF Paramete equency name for 1/pro- pisse file name imber of noise/freq paid C voltage	Edit Obje urrent Constant stem User Constant Reset Instant Reset Instant vpulse symbol v2 Add Master V3 TRUE eriod Instant Stem Constant Stem Constant Instant Stem Constant Stem Consta	Act Properties	Display off D off D off D off D Display off D off D off D off D off D)	
OK Cano	(a) (a) (b) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c	Utts Previous N pily To only c tow sys Property Library Name Cell Name View Name Instance Name User Property Ivsignore CDF Paramete equency name for 1/pro- pise file name imber of noise/freq pail C ontage C magnitude	Lexi Help	Act Properties	Display off D off D off D off D Display off D off D off D off D off D off D off D off D)	
Canc	eel Apply Defar (a) (a) (a) (a) (a) (a) (a) (b) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c	ults Previous N pply To only c pow sys Property Library Name Cell Name View Name Instance Name User Property Ivsignore CDF Paramete equency name for 1/pn bise file name imber of noise/freq pal C voltage magnitude phase	Lexi Help		Display off D off D Display off D Display off D off D off D off D off D off D off D)	
Canc	eel Apply Defar (a) (a) Ap Sh Sh Nu UC AC AC	ults Previous N pply To only c property Library Name Cell Name View Name Instance Name User Property Usignore CDF Paramete equency name for 1/p pise file name imber of noise/freq pai c voltage phase magnitude phase magnitude	text Help Edit Obje uurrent Imstan stem user ✓ Reset Instar Reset Instar malogLibl vpulse symbol v2 Add Master Va TRUE errod malogLibl v2 Add malogLibl v2 Add malogLibl v2 Add master Va TRUE	act Properties act Properties co CDF co Labels Display Value Delete Modify due Local Value Value Value	Display off D off D Display off D Display off D off D off D off D off D off D off D off D)	
OK Cano	eel Apply Defar (a) (a) Ap Sh Sh Fre No OC AC AC XF PA	ults Previous N pply To only c pow sys Property Library Name Cell Name View Name Instance Name User Property Ivsignore CDF Paramete equency name for 1/pr vise file name imber of noise/freq pai c onlage magnitude pase magnitude C ma	text Help Edit Obje turrent Instan tem user Reset Instar Reset Instar reset Instar vpulse symbol vz Add Master Va TRUE erriod Instan Insta	act Properties cce CDF cce Labels Display Value Delete Modify tue Local Value Value Value	Display off D Display off D Display off D Display off D Display off D off D off D off D off D off D off D off D)	
OK Cano	eel Apply Defar (a) (a) (a) (a) (a) (a) (a) (b) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c	Utts Previous N pily To only c tow system Property Library Name Cell Name View Name Instance Name User Property Ivsignore CDF Parameter CDF Parameter Source Parameter CDF Parameter CDF Parameter Source Parameter CDF Parameter CDF Parameter CDF Parameter CDF Parameter CDF Parameter CDF Parameter CDF Parameter CDF Parameter CDF Parameter Source Parameter CDF Parameter	Lexi Help	cc Properties ccc CDF cc LabelS Display Value Delete Modify Value	Display of D of D Display of D Display of D Display of D of D of D of D of D of D of D of D)	
OK Cano	eel Apply Defar (a) (a) (a) (a) (a) (a) (a) (b) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c	ults Previous N pply To only c poly To only c property Library Name Cell Name View Name Instance Name User Property Usignore CDF Paramete equency name for 1/p ise file name imber of noise/freq paid C voltage magnitude C magnitude C magnitude CC phase oitage 1	eed Help K Edit Obje iurrent ♥ Instan stem ♥ user ♥ Reset Instar vpulse syabol V2 Add Master Va TRUE er o 0 0 v 1 0 v 1 0 v 1 0 0 v 1 0 0 v 1 0 0 0 0 v 1 0 0 0 0 0 0 0 0 0 0 0 0 0	act Properties acce CDF CDF Value Use Value	Display of D Display of D Display of D Display of D Display of D of D of D of D of D of D of D of D)	
OK Cano	eel Apply Defar (a) (a) (b) (b) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c	In the second seco	Edit Obje Instan Stem ♥ user ♥ Reset Instar analogLib Vpulse syabol V2 Add Master Va TRUE er d 0 V 1.8 ♥ 4.8 ♥	act Properties ice ice CDF ice ice Value Delete Modify iue Value	Display of D Display of D Display of D Display of D Display of D Display of D of D of D of D of D of D of D of D)	
OK Cano	eel Apply Defar (a) (a) (b) (b) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c	Voltage Comagnitude	Leed Help	act Properties ccce cDF cccal Labels Display Value Delete Modify tue Value	Display off D Display off D Display off D Display off D Display off D off D of)	
OK Cano	eel Apply Defar (a) (a) (b) (b) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c	Utts Previous N pipiy To only c tow sys Property Library Name Cell Name View Name Instance Name User Property User Property User Property User Property Issignore CDF Paramete equency name for 1/pr pise file name imber of noise/freq pail C voltage C magnitude C phase r magnitude AC magnitude AC magnitude AC magnitude AC phase rind Ntage 2 rind Alaytime Set Imme	Leed Help	Act Properties	Display off D Display off D Display off D Display off D off)	
OK Cano	el Apply Defat (a) (a) (b) (b) (c) (c) (c) (c) (c) (c) (c) (c) (c) (c	View Name Instance Name View Name Instance Name View Name Instance Name User Property Library Name Cell Name View Name Instance Name ODF Paramete equency name for 1/p pise file name imber of noise/freq pai voltage Dase rangnitude Cphase Tangnitude Chase Vitage 1 pitage 1 pitage 2 rriod play time set time	Edit Obje urrent Instan Istem User Reset Instar analogLibl vpulse syabol v2 Add Master Va TRUE er or eriod In 0 0 V 1.8 v 4.01n s -100p s	cct Properties ccce ccb CDF cce cce	Display off D Display off D Display off D Display off D off)	

Figure 7.10: Sources in PFD+CP+Filter Testbench



Figure 7.11: PFD+CP+Filter Testbench

Now create a new schematic and name it 'Tb_vco'. This will be the testbench schematic from which we will run all our simulations to test that the VCO is functioning as expected. Insert 'vdc', 'gnd' and two 'vdc' from the Component Library by navigating to the 'Analog Parts' library. The initial conditions to be set for VDD are same as shown in Figure 7.10(a), while for the second 'vdc' source the DC voltage should be set to a parametric variable 'vctrl'. Figure 7.12 shows what your testbench schematic should look like at the end of this step.



Figure 7.12: VCO Testbench

Finally, create a new schematic and name it 'Tb_div'. This will be the testbench schematic from which we will run all our simulations to test that the VCO is functioning as expected. Insert 'vdc', 'gnd' and a 'vpulse' from the Component Library by navigating to the 'Analog Parts' library. The initial conditions to be set for VDD are same as shown in Figure 7.10(a), while for the second 'vpulse' source they are shown in Figure 7.13. Figure 7.14 shows what your testbench schematic should look like at the end of this step.
Browse	Reset Instance Labels Display	Distant
Property	Value	Display
Library Name	analogLib	off 🔽
Cell Name	vpulse	off
View Name	symbol	off 🔽
Instance Name	¥1	off 🔽
	Add Delete Modify	
User Property	Master Value Local Value	Display
Ivsignore	TRUE	off 🔽
CDF Parameter	Value	Display
Frequency name for 1/peri	bd	off 🧧
Noise file name		off 🧧
Number of noise/freq pairs	0	off 🧧
DC voltage	1.8 V	off 🔽
AC magnitude		off 🔽
AC phase		off 🔽
XF magnitude		off 🔽
PAC magnitude		off 🔽
PAC phase		off 🔽
Voltage 1	0 ¥	off 🔽
Voltage 2	1.8 ¥	off 🔽
Period	1/1.6e9 s	off 🔽
Delay time	0 s	off 🔽
Rise time	80p s	off 🔽
Coll time	80p s	off 🔽
r an ume		

Figure 7.13: Vpulse configuration for Divider



Figure 7.14: Divider Testbench

7.4.5 Circuit Simulation Using Spectre

7.4.6 Launching ADE

- 1. We will simulate our circuits using Cadence Spectre Simulation engine. Spectre is a variant of HSPICE developed by Cadence and provides greater accuracy, speed and flexibility especially when dealing with mixed signal circuits.
- 2. Make sure you first 'Check and Save' your test bench schematic and click on $Launch \rightarrow ADE$ to open up the ADE window as shown in Figure 7.15.
- 3. Click on $Setup \rightarrow Simulator$ to make sure the Simulator is set to Spectre as shown in Figure 7.15.



Figure 7.15: Simulating Circuit with ADE

4. Now click on Setup → Model Libraries to configure the Spectre model files. Figure 7.16 shows the path you need to browse to in order to get the correct model files for the PDK. You most likely would not need to manually type the model file paths as Virtuoso should take care of it, but in case you do the path is listed.

😑 🔿 🔿 🕅 🕅 🕅 🕅 🕅 🕅	etup
Model File	Section
🖻 Global Model Files	
	4
	2
	×
1	
	OK Cancel Apply Hel
(a)	
⊖ ○ ○ X spectre0: Model Library Setu	ıp
Model File	Section
Global Model Files July home/EEARPS/TSMC018/temc18rf/ /medals/cnectro/rf018.ccc	+
Violate/EEAPPS/TSMC018/tsmc18rf/./models/spectre/rf018.scs	tt 3v
— /home/EEAPPS/TSMC018/tsmc18rf//models/spectre/rf018.scs	tt_na 😱
// with the second s	tt_3vna
/nome/EEAPP3/T3MCUT6/tsmc18rf/_/models/spectre/rfu18.scs	<u>π_m</u>
/home/EEAPPS/TSMC018/tsmc18rf/ /models/spectre/rf018 scs	#_5/1
//home/EEAPPS/TSMC018/tsmc18nf//models/spectre/nf018.scs	tt_bip3
— /home/EEAPPS/TSMC018/tsmc18rf//models/spectre/rf018.scs	dio 🗙
//home/EEAPPS/TSMC018/tsmc18rf//models/spectre/rf018.scs	dio3
//nome/EEAPP3/T3MC016/tsmc18rf/ /models/spectre/rf016.scs	dio_anw # rec
//initial/EEAPPS/TSMC018/tsmc18rf/ /models/spectre/rf018.scs	#_105
//home/EEAPPS/TSMC018/tsmc18rf//models/spectre/rf018.scs	tt rfmos
— /home/EEAPPS/TSMC018/tsmc18rf//models/spectre/rf018.scs	tt_rfmos33
— /home/EEAPPS/TSMC018/tsmc18rf//models/spectre/rf018.scs	tt_rfmim
//home/EEAPPS/TSMC018/tsmc18rf//models/spectre/rf018.scs	tt_rfind
/nome/EEAPPS/TSMC018/tsmc18ff./models/spectre/ff018.scs	tt_rtmvar
✓ /nume/EEAPP3/13/VICUT0/tsmic1on//mudels/spectre/mut8.scs	IL_FIJV&F
///initial/contents/spectre/f010.scs	tt rfres rno
/home/EEAPPS/TSMC018/tsmc18rf//models/spectre/rf018.scs	tt rfres hri
//home/EEAPPS/TSMC018/tsmc18rf//models/spectre/rf018.scs	tt_rfesd
— /home/EEAPPS/TSMC018/tsmc18rf//models/spectre/rf018.scs	tt_bbmvar
	OK Cancel Anniu Hal
	Cancer Chipping Her
<i>(-</i>)	
(b)	

Figure 7.16: Configuring Model Files

7.4.7 PFD+CP+Filter ADE Setup

1. Transient analysis of any circuit is key to study the time domain behavior. We will simulate the PFD+CP+Filter testbench and observe the resulting plots for 'UP', 'DN', signals to ensure that the PFD is functioning properly.

- 2. In ADE window click on the **AC**,**DC**,**Tran** icon on the right pane. Choose the 'tran' simulation type, pick the stop time to be 100ns and choose 'moderate' in the 'Accuracy details'.
- 3. Click on Variables $\rightarrow Copy \ From \ Cellview$ and insert the filter parameters as shown in Figure 7.17.
- 4. Click on the green 'Play' button to run the simulation and the plots should automatically pop up in a new output window. If you right click on the name of the signal listed in the left panel, you can navigate to options that change the thickness and color of the output waveform. Additionally, right-clicking anywhere on the output window and navigating to 'Graph Properties' allows you to alter the background color as well.



Figure 7.17: PFD+CP+Loop-Filter Testbench ADE Window

5. Your final output waveform should look like that shown in Figure 7.18. Notice that the UP,DN pulses are appropriately modulated as 'REF' and 'DIV' signals diverge from one another; thus, the PFD is indeed functioning correctly.



Figure 7.18: PFD Spectre Simulation Output

7.4.8 VCO ADE Setup

Recall from the earlier sections that for oscillators it is critical to perform Transient as well as PSS simulations. We will therefore simulate the VCO testbench and observe the resulting transient and PSS simulation outputs for 'Vout' signals to ensure that the VCO is functioning properly. Additionally, since the VCO is the major noise contributor to the PLL, we will also characterize the VCO Phase-Noise by performing PNoise simulation on the testbench.

- 1. In ADE window click on the **AC**,**DC**,**Tran** icon on the right pane. Choose the 'tran' simulation type, pick the stop time to be $4\mu s$ and choose 'moderate' in the 'Accuracy details'.
- 2. Click on Variables \rightarrow Copy From Cellview and insert the PMOS, NMOS width and 'vctrl' parameters as shown in Figure 7.19.
- 3. Click on the green 'Play' button to run the simulation, and the plots should automatically pop up in a new output window. If you right-click on the name of the signal listed in the left panel, you can navigate to options that change the thickness and color of the output waveform. Additionally, right-clicking anywhere on the output window and navigating to 'Graph Properties' allows you to alter the background color as well.



Figure 7.19: VCO Testbench ADE Window

4. Your final output waveform should look like that shown in Figure 7.20. Notice that the output node voltage 'vout' is oscillating thus the VCO is indeed functioning correctly.



Figure 7.20: VCO Transient Simulation Output

5. For a VCO, a key figure-of-merit is the control voltage tuning range. Thus, we have to perform a parametric analysis in order to observe the change in 'frequency' as well as K_{VCO} as a function of 'Vctrl'. In order to do so in the ADE setup window click on $Tools \rightarrow Parametric$ Analysis and a window like Figure 7.21 should pop-up. Within the parametric analysis window, when you double-click on the variable box, a drop-down list will show up from which you should pick 'vctrl'.



Figure 7.21: Vctrl Parametric Analysis Setup

To run the parametric analysis, click on the 'Play' within the Parametric-Analysis window. This setup is basically going to run the transient simulation $\frac{To-From}{StepSize}$ times by varying the control-voltage input to the VCO.

6. To plot frequency vs. Vctrl and K_{VCO} vs. Vctrl we need to use the 'Calculator' tool in-built within ADE. Click on $Tools \rightarrow Calculator$. The Calculator window as shown in Figure 7.22 will open up and within it now you should select 'Vt' from the toolbar. The schematic will open up, so within the schematic select the 'vout' node. From the 'Function-Panel' within the Calculator window choose the 'frequency' and 'average' functions to make up the function shown in Figure 7.22. Now go back to the ADE window, click on the right-pane and select the 'Pick-Outputs' button. A window will pop up so within it select 'Get-Expression' and name it 'freq'. This will bring the expression you just created in the Calculator so that you can plot it. Conversely, you can also click on the 'plot' button shown in the red-box in Figure 7.22 to plot the expression; however, doing so makes the title of plot look a little too crammed.

000)		2	Virtuos	o (R) Visualia	zation & Ana	alysis XL calcu	lator				
<u>F</u> ile <u>T</u> o	ools <u>V</u> iev	/ <u>O</u> ptions <u>C</u> on:	stants <u>H</u> e	lp							cād	ence
In In	Context R	esults DB: none	specifed									
III app	plot erp	lot										
O vt	🔾 vf 📗	⊖ vdc ⊖ vs	🔾 op	🔾 var	○ vn ○) sp 🔾 vsv	vr 🔾 hp 🔾	zm				
0 it	⊖ if	◯ idc │ ◯ is	🔾 🔾 opt	🔾 mp	○ vn2 ○	zp 🔾 yp	🔾 gd 🔾	data				
le or (🔾 Family	🔾 Wave 🛛 🗹 CI	ip 🏹	New New	Subwindow	Rectangu	lar 🔽 🎊	8				
Key	ð×	average (freq	VT("/vout"	") "rising" ?:	<name "time"<="" td=""><td>?mode "auto</td><td>" ?threshold 0.</td><td>3)<mark>)</mark></td><td>_</td><td>_</td><td>_</td><td></td></name>	?mode "auto	" ?threshold 0.	3) <mark>)</mark>	_	_	_	
7 8	9 /											
4 5	6 ×											
		5										
			Pop		Here Mare I	expr expr	fn fn					
0 ±	. +		insert	ا شا		11 P)				
Stack												ē×
o eye	eDiagram(v("/Voutn" ?result	t "tran") 3u	3.05u 1/1.	6G)							
100 VC	(Vouto" 2t	esult "tran")										
	rvouir si	count alony										
1.66	63E-6											
💿 ave	erage(clip)	(i("/17/115/currento	ut" ?result	"tran") Ou	1.2u))							

Euroction	n Panol											a y
Tuncuu	ii i dulei											
MI			<									
1/x 10**×	acos acosh	b1f bandwidth	convolve	dBm delav	evmQAM evmQnsk	freq freq iitter	gainMargin getAsciiWave	gpc_gain groupDelay	ifreq ih	ipn ipnVRI	lsb Ishift	oversho
PN	asin	clip	cosh	deriv	exp	frequency	gmax	gt	iinteg	itime	mag	peak
Rn a2d	asinh	compare	cross d2a	dft dfthh	eyeDiagram fallTime	ga gac freg	gmin	gumx	imag int	kf	nc_freq	peakToF
abs	atanh	compressionVRI	dB10	dnl	flip	gac_neq gac_aain	amoa	harmonicFreq	intea	loadpull	nc_yam	phase
abs_jitter	average	conjugate	dB20	dutyCycle	fourEval	gainBwProd	gpc_freq	histo	intersect	log10	nfmin	phaseDr
17												
17												1

Figure 7.22: ADE Calculator

7. Repeat the same steps as above to create an expression within the calculator to compute the K_{VCO} . Use the 'deriv' function within the Calculator Function Panel to do so. Finally, click on the 'Play' button within the ADE window to plot frequency vs. Vctrl and K_{VCO} vs. Vctrl curves. Your output should look like Figure 7.23.



Figure 7.23: Frequency vs. Vctrl and K_{VCO} vs. Vctrl Simulation Plots

The K_{VCO} calculated in Figure 7.23 is the value used to calculate the required charge-pump current as well as C_1 and C_2 values from the loop-filter. In our case, the VCO output frequency is 1.675GHz with a Vctrl=0.9V and a $K_{VCO} = 652.106$ MHz/V.

- 8. Now, in order to simulate the VCO Phase-Noise we need to perform the PSS, PNoise simulations. For the PSS simulation, in the ADE window click on the **AC,DC,Tran** icon on the right pane. Choose the 'pss' simulation type, pick the parameters using those shown in Figure 7.24(a). It is critical to note that the beat frequency here is the target frequency of the VCO and the reason we have to check the 'oscillator' option and select 'vout', 'gnd' terminals from the schematic is because by default PSS simulation expects a differential output.
- 9. To run the PNoise simulation, in the ADE window click on the **AC,DC**, **Tran** icon on the right pane. Choose the 'pss' simulation type, pick the parameters using those shown in Figure 7.24(b). Its critical to note that the phase-noise in the VCO is only dominant in the low-pass thus we limit our simulation frequency range to be from 1kHz to 10MHz as after the 10MHz the phase-noise will not cause any significant degradation to oscillator output performance. <u>Note</u>: To view the PNoise simulation results in the main ADE window click on *Results* \rightarrow *Direct Plot* \rightarrow *Main Form* at which a window like Figure 7.25. Choose 'Phase-Noise' and click on the 'Plot' button.

● ○ ○ X	Choosing Analyses Virtuos	® Analog Desig	
Analysis	🔾 tran 🛛 dc 🔾 ac	🔾 noise	
	🔾 xf 💦 🔾 sens 🔾 dcmatch	🔾 stb	
	🔾 pz 🛛 🔾 sp 🔾 envlp	💌 pss	
	🔾 pac 🛛 🔾 pstb 🔾 pnoise	⊖ p×f	😑 🔿 🔿 🔀 Choosing Analyses Virtuoso® Analog Desig
	🔾 psp 🛛 🔾 qpss 🔾 qpac	o qpnoise	Periodic Noise Analysis
	🔾 qpxf 🛛 🔾 qpsp 🔾 hb	🔾 hbac	PSS Beat Frequency (Hz)
	🔾 hbnoise 🔾 hbsp		
	Periodic Steady State Analys	is	Multiple phoise
Engine	🖲 Shooting 🔾 Harmonic Bala	nce	
			Sweeptype default Relative Harmonic
			Output Frequency Sweep Range (Hz)
Fundame	Fror Value Sign	al SreId	Chart Stan
W Moule	Evbi varue Sign	ar sicia	Start-Stop Start 1k Stop 100e6
1 vout	1.6e9 1.66 Larg		Sweep Type
			Logarithmic Points Per Decade 1601
I			Number of Steps
vout	1.6e9 1.60 Large		Add Specific Points
Clear/A	Add Delete Update From	n Hierarchy	
			Sidebands
🖲 Beat F	requency 1.60	Auto Calculate 🗔	Method 💿 default 🔾 fullspectrum
🔾 Beat F	Period		Maximum sideband
Number of	rmonics		when using shooting engine, default value is 7.
(Tumber o			Output
			Positive Output Node /vout Select
Accuracy	Defaults (errpreset)		Negative Output Node / and Select
⊻ conse	ervative 🗌 moderate 📃 liberal		
Additional	Time for Stabilization (tstab)		Input Source
Save Initial	Transient Results (saveinit) 🛛 🗌	no 🗆 yes	none
Oscillator	✓ Oscillator node+ /vout	Select	Noise Type sources
	Oscillator node- /gnd!	Select	sources: single sideband (SSB) noise analysis
🖌 Calcul	late initial conditions (ic) automatica	Iv	
		·	concrete poise into course and gain
Sweep			separate noise mu source and gain
	OK Cancel De	faults Apply Help	OK Cancel Defaults Apply Hel
	(a)		(b)
	(a)		(D)

Figure 7.24: VCO PSS & Pnoise Simulation Setup

000	C Direct Plot Form
Plotting Mode	Append
🔾 pss 💿 pnois	e .
Function	
Output Noise	 Input Noise
🔾 Noise Figure	 Noise Factor
NFdsb	⊖ Fdsb
NFieee	⊖ Fieee
Phase Noise	Transfer Function
Loadpull Contour	· 🗆
Add To Outputs	Plot
> Press plot butto	on on this form
	OK Cancel Help

Figure 7.25: VCO Phase-Noise Simulation Plot Step

The output waveform for the simulated phase-noise will look like Figure 7.26. In our case, we find that the Phase-Noise at a 1MHz offset is equal to -94.32dBc/Hz, which is very reasonable for a ring-oscillator type single-ended VCO topology.



Figure 7.26: VCO Phase-Noise Simulation Plot

7.4.9 Divider ADE Setup

- 1. We now simulate the divider testbench and observe the resulting plots for 'IN', 'OUT', signals of the divider circuit to ensure its proper functionality.
- 2. In ADE window click on the **AC**,**DC**,**Tran** icon on the right pane. Choose the 'tran' simulation type, pick the stop time to be 100ns and choose 'moderate' in the 'Accuracy details'.
- 3. Click on the green 'Play' button to run the simulation and the plots should automatically pop up in a new output window.



Figure 7.27: Divider Spectre Simulation Output

4. Your final output waveform should resemble Figure 7.27. Notice that each half-wave of the output pulse comprises of four half-pulses of the input, meaning the period of the output pulse is one-eighth of the input pulse period. Thus, our divider is functioning properly in that it divides the input pulse frequency by 8 with a small setup-time delay of 209ps.

7.4.10 Complete PLL Schematic and Testbench

- 1. Using the steps mentioned in the subsections above create new schematic and save it as 'PLL'.
- 2. Place the 'pfd', 'vco' and 'div' symbols in the schematic, connect the components in together and generate a symbol for the full PLL schematic as shown in Figure 7.28.



Figure 7.28: PLL Schematic

3. Create a new schematic and save it as 'Tb_PLL'. Design the testbench schematic as shown in Figure 7.29.

	•			•	·											•						•	·			•		•					ŀ	
	Γ																•]									•	•	-	Ś		
																				÷														
0	ť	Z Vdi	 1.8																ę	3.														
2	ŕ					·	•		•	٦	ef	•	- <u>-</u>		Fre		be	t	tre		DLI			·	•		•							
	I					Ċ														рш 2.				:										
						l.,													.7	5.														
					пČ	b	v1:2 v2=	1.8												•														
					•	ľ	tr-	100	P										Ì															
						ŀ																												
						•																												
					_														_															
																				5														

Figure 7.29: PLL Testbench

4. In the PLL testbench choose the 'Vref' using the 'Vpulse' source within analogLib and configure it as shown below in Figure 7.30.

Browse	Reset Instance Labels Display	
Property	Value	Display
Library Name	analogLib	off 🔽
Cell Name	vpulse	off 🔽
View Name	symbol	off 🔽
Instance Name	₩1	off 🔽
	Add Delete Modify	\supset
User Property	Master Value Local Value	Display
lvslgnore	TRUE	off 🔽
CDF Parameter	Value	Display
Frequency name for 1/perio	d	off 🔽
Noise file name		off 🔽
Number of noise/freq pairs	0	off 🔽
DC voltage	1.8 ¥	off 🔽
AC magnitude		off 🔽
AC phase		off 🔽
XF magnitude		off 🔽
PAC magnitude		off 🔽
PAC phase		off 🔽
Voltage 1	0 V	off 🔽
Voltage 2	1.8 ¥	off 🔽
Period	5n s	off 🔽
Delay time	100p s	off 🔽
Rise time	100p s	off 🔽
Fall time	100p s	off 🔽
	0.4	off

Figure 7.30: Vpulse Configuration for PLL Testbench

7.4.11 PLL ADE Setup

Now that we have verified the functionality of each of the components of the PLL at a transistor level, the final task is to characterize the PLL locking behavior, overall phase-noise profile and jitter profile.

- 1. In ADE window click on the **AC**,**DC**,**Tran** icon on the right pane. Choose the 'tran' simulation type, pick the stop time to be $4\mu s$ and choose 'moderate' in the 'Accuracy details'.
- 2. Click on Variables \rightarrow Copy From Cellview and insert the PMOS, NMOS widths of VCO inverters, charge-pump output current, I_{CP} , reference signal period, and the loop-filter parameters as shown in Figure 7.31.

3. Click on the green 'Play' button to run the simulation and the plots should automatically pop-up in a new output window.



Figure 7.31: PLL Testbench ADE Window

4. Your final output waveform should look like that shown in Figure 7.32. Notice that the VCO input control voltage 'vctrl' is essentially flat and settled thus the PLL is in steady-state lock state.



Figure 7.32: PLL Settled Transient Simulation Output

If we zoom into a 50ns window we notice that there is a slight control voltage ripple, but the loop is approaching steady state lock point. From Figure 7.33 the rippling behavior of 'vctrl' can be seen to be prominent for the first 10ns and then slowly decaying away as we approach 50ns time-frame.



Figure 7.33: PLL Vctrl Voltage before Lock

5. Another method to verify that PLL is in steady-state locked condition is to plot the output frequency versus time. Figure 7.34 shows that the PLL achieves lock around $1\mu s$ and remains locked to 1.6GHz output frequency after that.



Figure 7.34: PLL Frequency Locking in Steady-State

To plot frequency we need to use the 'Calculator' tool in-built within ADE. Click on $Tools \rightarrow Calculator$. The Calculator window as shown in Figure 7.22 will open up and within it now you should select 'Vt' from the toolbar. The schematic will open up, so within the schematic select the 'vout' node. From the 'Function-Panel' within the Calculator window choose the 'freq' function to plot the PLL output frequency with respect to time. Once again, like in the case of the VCO frequency vs. vctrl plot, go back to the ADE window, click on the right-pane and select the 'Pick-Outputs' button. A window will pop-up so within it select 'Get-Expression' and name it 'freq'. This will bring the expression

you just created in the Calculator so that you can plot it.

- 6. Now, in order to simulate the PLL Phase-Noise we perform the PSS, PNoise simulations. For the PSS simulation, in the ADE window click on the AC,DC,Tran icon on the right pane. Choose the 'pss' simulation type and pick the parameters using those shown in Figure 7.35(a). Note that in this case the beat frequency will be the reference frequency as that is the only fundamental input frequency to the PLL. Additionally, the reason we do not have to check the 'oscillator' option and select 'vout', 'gnd' terminals from the schematic is that PLL is, as the name suggests, not an oscillator.
- 7. To run the PNoise simulation, in the ADE window click on the AC,DC, Tran icon on the right pane. Choose the 'pss' simulation type, pick the parameters using those shown in Figure 7.35(b). It is critical to note that the phase-noise in the VCO is the dominant source of phase-noise in the complete PLL, and since VCO noise is typically most prominent at a 1MHz offset, we limit our simulation frequency range to be from 1kHz to 10MHz as after the 10MHz the phase-noise will not cause any significant degradation to oscillator output performance. One key difference between the PNoise setup and VCO is that now the phase-noise of interest is of the 8th relative harmonic to the fundamental reference frequency because we have a divider ratio of 8 in our PLL.



Figure 7.35: PLL PSS & Pnoise Simulation Setup

The output waveform for the simulated phase-noise will look like Figure 7.36. In our case, we find that the Phase-Noise at a 1MHz offset is equal to -113.31dBc/Hz, which is very reasonable for an Integer-N clock synthesizer PLL with an output frequency of 1.6GHz in steady-state.



Figure 7.36: PLL Phase-Noise Simulation Plot

8. Since clock generator circuits are responsible for generating the system master-clock, the timing (deterministic) jitter as well as random jitter are key figures-of-merit to minimize clock-induced timing errors during transmission as well as reception of digital data bits. In order to characterize the deterministic timing jitter we plot the PLL eye diagram for a small time-interval once the PLL is in lock condition. Similar to the frequency measurement, we plot the eye diagram by exporting the 'vout' curve into Calculator and using the 'eyeDiagram' function as shown in Figure 7.37.

😑 🔿 🔿 📉 Virtuoso (R) Visualization & Analysis XL calo	ulator
<u>Eile Tools View Options Constants H</u> elp	cādence
In Context Results DB: /home/rishi/simulation/boottrapPLL_test/spectre/schematic/psf	
III app plot erplot	
Ovt Ovf Ovdc Ovs Opp Ovar Ovn Osp Ovswr Ohp O	zm
llloit Oif Oidc Ois Oopt Omp Ovn2 Ozp Oyp Ogd O	data
📗 Off 🔾 Family 🔾 Wave 📝 Clip 🍢 🐗 New Subwindow 🥊 Rectangular 🔽 🥸	
Key 🗗 🗶 eyeDiagram v("/Voutn" ?result "tran") 3u 3.05u 1/1.6G	
7 8 9 /	
4 5 6 *	
123-	
□ ± . + + ┛ ₽ ₩ ₩ ₽ ₽ ₩ ₩ ₩ ₩ ₩ ₩ ₩	9 ¢

Figure 7.37: Eye Diagram Setup for PLL Output

In Figure 7.38 we see that the output voltage eye for the PLL has some deterministic timing jitter associated with it.



Figure 7.38: PLL Output Eye Diagram Plot

If we zoom-in to the plot, in the area of the 'vout' rising-edge, and place markers at points where the voltage passes 1.0V, we notice that the deterministic timing jitter of our PLL as shown in Figure 7.39 is 5.62ps.



Figure 7.39: PLL Deterministic Jitter Plot

Finally, to calculate the random edge-to-edge jitter of our PLL we need to re-run the PNoise simulation but this time with jitter. In order to do so change the 'Noise-type' in the PNoise setup shown in Figure 7.35(b) to 'jitter'. Once you have run the PNoise simulation with jitter, navigate to Results \rightarrow Direct Plot \rightarrow Main Form from the main ADE window and a window like Figure 7.40 will pop up. Choose 'Jee' and pick an event-time at a point where PLL is in steadystate lock. You can also choose a specified BER. In modern links the BER is typically 10⁻¹², but we plot the jitter over various BER ranges as shown in Figure 7.41.

Direct Plot Form (on natcsi.ece.illinois.edu)	×
Analysis	\sim
o pss o tdnoise	
• pnoise jitter	
Function	
⊖ Threshold Xing ● Jee	
⊖ Jc ⊖ Jcc	
Event Time 2.50603n	
Signal Level 🔾 rws 🖲 peak-to-peak	
BER 1e-12	
Modifier	=
Integration Limits	
Start Frequency (Hz 1K	
Stop Frequency (Hz) 1M	
edd To Outputs	
> Press plot button on this form	
OK Cancel H	alp

Figure 7.40: PLL Random Jitter Plot Setup



Figure 7.41: PLL Random Jitter Plot

This concludes the characterization of the clock-generator circuit at a transistor level and we have successfully verified its proper functionality.

CHAPTER 8

DISCUSSION

8.1 Conclusion

Overall, thus far in this thesis the foundational motivation for the use of High-Speed Serial Links has been described, fundamentals of PLLs as well as Charge-Pump PLLs have been covered, and an in-depth step-by-step tutorial on design/simulation of an on-chip clock synthesizer at both a behavioral level using Verilog-AMS and transistor level using Cadence Spectre have been described. The designed PLL based clock-generator circuit described in Chapter 5 operates at an output frequency of 1.6GHz at lock with -113dBc/Hz phase-noise, 5.62ps deterministic jitter and 5.27ps edge-to-edge random jitter at a BER level of 10^{-12} . Although many design improvements can be made at the circuit level to optimize the phase noise and jitter performance of the clock-generator circuit, the motivation for this thesis was to provide a tutorial style training manual for a student pursuing mixed-signal IC design at the beginning of their graduate studies; thus, the circuits used for the Integer-N synthesizer are very basic/standard. In this last chapter to conclude the thesis, a summary of future design improvements for the circuit designed/simulated in this thesis is outlined from both a system as well as circuit architecture level. Lastly, an outline of the potential areas of research to explore in the field of high-speed serial links design with a signal integrity focus is presented. Often times when pursuing graduate work in a diverse and mature field of Electrical and Computer Engineering such as Mixed-Signal Circuit design, especially with a focus on Signal Integrity, a new student needs some guidance and initial training to jump-start their careers. Therefore, the final section of this thesis concludes with a few words of advice for new students pursuing this field of study to enable them in solving unexplored areas within this field.

8.2 Future Work

8.2.1 Design Improvements

The PLL based clock-generator circuit designed and simulated in this thesis is very basic and is not optimized for optimal phase-noise and jitter performance for use as an on-chip clock signal in modern day high-speed serial links. Such a simple circuit topology was however chosen in order to simplify the complexity of the system and allow non-circuit designers, especially engineers in the field of signal integrity, to understand the basics of the mixed-signal circuit design/simulation flow using the ubiquitous tool like Cadence Virtuoso. The following list outlines some of the design changes that enhance the speed as well as robustness by optimizing the phase-noise, jitter and power consumption of the on-chip synthesizer:

- 1. Replace the NAND-PFD with other PFD topologies such as Pass-Transistor or Glitch-Latch Flip-Flop.
- 2. Implement an actual biasing current-sink in the Charge-Pump and drive this sink by an external off-chip voltage signal. Additionally, implement a differential charge-pump to minimize the CP induced noise and UP/DN current mismatch.
- 3. If using a ring-oscillator VCO topology, bias the circuit using a selfbiasing circuit to reduce the control voltage ripples as well as suppress the supply-induced noise. Additionally, the oscillator should be made differential instead of single-ended as this would greatly improve the phase-noise and jitter performance [9].
- 4. Implement a TSPC Split-Output latch within the divider or a modified variant of it to reduce the divider-induced delay which translates into additional jitter at the output.
- 5. Consider implementing a LC-tank based VCO. Although it will increase the area of the circuit, the phase noise and jitter performance will be much better than even a differential ring-VCO design [10, 1].
- 6. Perform Monte Carlo analysis within Cadence ADE XL to optimize the design over PVT corners to select the optimal sizing for the transistors.

7. The industry-wide trend is to move towards all-digital PLLs. Digital PLLs offer many advantages over analog PLLs mainly in the fact that they eliminate the need for an analog loop-filter as well as a charge-pump, thereby saving area and power consumption. The disadvantage currently however is that quantization noise associated with the Time-to-Digital Converter (TDC) inside a digital PLL severely degrades the phase noise of the system. Thus, even the current state-of-the-art DPLLs are no match in terms of spectral purity performance at high-speeds compared to the analog PLLs. However, with improvements in transistor scaling and machine learning algorithmic noise-tolerance methodologies in digital signal processing, as well as stochastically enhanced circuit design techniques, the future really lies in the study of synthesizable DPLLs. Thus, this area of study should definitely be of utmost priority while performing research on clocking circuit design for high-speed serial link applications [11, 12].

8.2.2 Signal Integrity Focus

In the realm of signal integrity (SI) engineering, the principle of utmost importance is to ensure robust signaling between a driver transmitter and a receiver across a channel medium. As clock frequencies and the associated data rates keep rising with technology scaling and advances in SFT while the channel bandwidth remains roughly the same and package sizes rapidly scale down, the need for superior SI designs in high-speed digital systems is at an all-time high with demand only increasing in the upcoming years. Since the off-chip I/O BWs become the major design bottleneck while demand for low-power designs becomes ubiquitous, it is critical to study SI problems in high-speed serial links.

Typical high-speed serial links are limited by the electrical PCB channels; thus, being able to model all the effects of the channel is key to designing robust systems. The skin-effect has become a major problem in PCB trace channels as the high-frequency signals experience a large series resistance due to current migration toward conductor outer surfaces, and traditional modeling techniques are no longer useful as they assume the metal surfaces are perfectly smooth, whereas in reality there is a significant roughness present which is actually random in nature. Furthermore, the frequency-dependent dielectric-losses experienced along these PCB substrates are difficult to characterize due to lack of the required sophistication in measurement techniques as well as statistical modeling. Measurement is a challenge because for accurate models the setup needs to be passive as well as causal, both of which are very difficult to ensure in practice. Therefore, statistical modeling of the substrate effects would be extremely valuable to enhance the understanding of the channels used in high-speed interface systems. Recall that a backplane or PCB trace can essentially be treated as a transmission line; thus, being able to quantitatively characterize propagation delay, system characteristic impedance, as well as discontinuities during high speed signaling will be very valuable in easing out equalization requirements on both TX as well as RX sides [13].

Overall link performance is analyzed in terms of the TX/RX timing jitter as well as BER specifications. Since the channel is fixed, as mentioned earlier, one prominent method to combat the jitter and ISI-inducing effects of the channel is to perform equalization. Sophisticated TX pre-emphasis equalization and RX side adaptive DFE are vital to link designers as the data rates approach the Tb/s ranges over the upcoming years while the off-chip I/O BW remains about the same. Lastly, formulation of a robust statistical BER analysis and time-domain empirical jittery analysis framework involving all interference sources for any given HSSL will be key to fully characterizing the non-idealities present in links today without the need to over/under design at a circuit level.

8.2.3 Steps for New Students

When embarking on a journey to study and solve new problems in the field of High-Speed Serial Link design, whether it be specifically in the area of clocking circuits, equalization or signal integrity, it is critical to have strong system-level understanding of the HSSL architecture. Once familiarized with the system level basics of links, the student (if interested in circuit design) should take ownership of a specific block within the overall link and focus on optimizing it for low-power, high-speed applications. Conversely, students interested in exploring signal integrity issues in HSSLs should familiarize themselves with Behavioral modeling as well as basics of transistor-level design and simulation analysis so that they explore new HSSL architectures that have high signal integrity even at multi-GHz to THz speeds.

Ideally, in order to make meaningful contributions in the field of HSSL designs, a strong knowledge-base in fields of Integrated Circuits, Electromagnetics, RF/Microwave theory and Digital Signal Processing is key. Thus, at the onset of their graduate career students performing research in the area of robust, fast-signaling HSSLs should take the fundamental graduate-level courses in areas of Digital IC Design, Analog IC Design, Phase-Locked Loop Design, Electromagnetics and DSP. Lastly, ability to exercise the EDA tools like Cadence Virtuoso, Agilent ADS, Ansys HFSS as well as programming in MATLAB and Verilog are essential in order to gain hands-on experience and perform rapid-prototyping of new research ideas.

APPENDIX A

CADENCE VIRTUOSO INSTALLATION GUIDE

A.1 Introduction

The motivation for this manual is to provide a step-by-step tutorial on installing Cadence Virtuoso IC 6.15 tools from scratch, configuring the environment and using the tool to design and simulate circuits. In this shorttutorial users are exposed to the complete steps involved in configuring their machine to run the Cadence Virtuoso IC 6.15 design environment along with its ancillary softwares, converting their host computer into a server, remotely connecting to it and launching the Virtuoso simulator engine from the terminal window followed by a detailed guide to create their own custom circuits and simulate them using the Cadence Spectre circuit simulator.

Cadence is an Electronic Design Automation (EDA) environment that integrates various circuit design and verification applications and tools (both in-house proprietary as well as external third party vendor tools) in a single framework allowing unified IC design and verification in a single environment. The tools are generic and allow the designer to configure the environment depending on the fabrication technology of choice by installing the appropriate PDK (Process-Design Kit).

This tutorial document is not intended to be a one-stop reference for all the features available in Cadence Virtuoso Design Environment. Instead, it is only meant to be a quick-start guide for circuit designers to be able to use the EDA tool to effectively simulate their designs for quick prototyping and verification of their designs.

A.2 Environment Setup

A.2.1 Installing Cadence Virtuoso

- Cadence Virtuoso design tools only work on Linux OS and best on RedHat based systems. In the scientific community a stable OS capable of running Cadence well is Scientific Linux which is an opensource Linux OS inspired from RedHat Enterprise Linux OS. Install the SL 6.4 64-bit x86 version from [http://ftp1.scientificlinux. org/linux/scientific/6.4/x86_64/iso/]. Make sure you install the SL-64-x86_64-2013-03-21-Everything-DVD1.iso and SL-64-x86_64-2013-03-21-Everything-DVD2.iso as it is the full enterprise version. Note: If your machine is 32-bit you can also install the 32-bit version from [http://ftp1.scientificlinux.org/linux/scientific/6.5/ i386/iso/http://ftp1.scientificlinux.org/linux/scientific/6.5/
- 2. Once you have installed the OS, make a new directory under the path /home/EEAPPS and name it CADENCE_INSTALL. In this tutorial we will be installing 'Virtuoso IC6.15' suite, 'MMSIM 11.1' (required for Spectre/Spectre-RF simulators), 'IUS8' (used for Verilog simulations in Cadence Design Suite), 'HSPICE' and a few 'PDKs' (Process Design Kit). Download the Cadence Virtuoso IC6.15 files (from your ftp file sever) and store them in your computer under /home/EEAPPS/CA-DENCE_INSTALL/IC615 folder. This folder is the location where you will keep the raw installation files during installation.

<u>Note:</u> When you are downloading your Cadence Virtuoso files from your ftp server location they will most likely be in *tar* file formats. You will have 7 files for Base version and 8 files for the Hotfix version. First download them into your Downloads folder and then extract the files one by one into the */home/EEAPPS/CADENCE_INSTALL*. Make sure while you *untar* your files you 'untar' each 'Base' file into the same folder and each 'Hotfix' file into the same folder so at the end of the whole process you will have two folders inside the 'CA-DENCE_INSTALL' folder named *IC06.15.011_lnx86.Base* and *IC06.15.132-615_lnx86.Hotfix*. Although all the installation will be performed from the Hotfix files it is **very important** to also have the Base files extracted as a path to them will be needed during the installation setup.

- 3. Before you start the installation process open up a terminal window and type in su to make sure that you have **root** user privileges.
- 4. A key feature of Scientific Linux environment to note is that if you ever have any missing packages that cause an error you just have to type in **yum install** *PackageName* in the terminal window. We will be using this throughout the installation process when we encounter such situations.
- 5. Install the following packages:
 - (a) yum install elfutils elfutils-libelf libXp
 - (b) yum install libXext.i686
 - (c) yum install libelf.so.1
 - (d) yum install libXrender.so.1
 <u>Note:</u> You need these packages for InstallScape (Cadence Installation Wizard) to work.

6. Create a new directory by typing: *mkdir -p /home/EEAPPS/CADENCE_INSTALL/IC615/*. Now move

the extracted Base and Hotfix folders to the 'IC6.15' folder you just created.

7. In the terminal window browse to the following folder: cd /home/EEAPP-S/CADENCE_INSTALL/IC615/ IC06.15.132-615_lnx86.Hotfix/CDROM1 and then type in sh SETUP.SH to start the installation process [14].

t InstallScape(TM) Classic View	_ = ×
InstallScape(TM) Classic View	cādence
Eile <u>V</u> iew <u>P</u> references <u>H</u> elp	
View installed Configure Update Uninstall Search & Install Control file Local directory/ Remove Favo	orite
releases releases releases releases install Media install Downloads Releases	ases
Local directory/media install	
	Provide Feedback
	Privacy Policy
Filter releases by: Select location of media Select location of media	
O Family	
ALL (1) IC615 (1)	
(ver/lace)/CADENCE INFTAIL/IC61/C061E122_61E_by/6DB0M1	
Type in location of media directory or click on Browse to select a media directory	Browse
<u>Exit</u>	Continue>

Figure A.1: Installing Virtuoso from Installscape Step 1

- 8. You will now see the following instructions so follow the steps indicated below very carefully:
 - (a) Specify path of install directory [OR type [RETURN] to exit]: /home/EEAPPS/IC615
 - (b) Directory /home/EEAPPS/IC615 does not exist. Create? [y/n]:
 y
 - (c) Do you have InstallScape for lnx86 platform installed somewhere [y/n]? **n**
 - (d) Do you want to install InstallScape for lnx86 [y/n]? y
 - (e) Type the path to InstallScape installation directory [(q to exit)]: /home/Cadence/InstallScape
- 9. Now a window like Figure A.1 will pop up so follow the instructions shown in it to browse into the correct folder that contains the installation files.

- 10. From this point onwards follow the instructions shown in Figures A.2 through A.14 very carefully to complete the installation process for Virtuoso. Make sure you do exactly as shown in these figures to ensure the software gets installed properly.
- 11. Once you reach the last step as shown in Figure A.14 hit 'Done'.

)	InstallScape(TM) Classic View	_ 0
InstallScape(TM) Classi	c View	cādence
<u>File V</u> iew <u>P</u> references <u>H</u> elp		
View installed Configure releases releases	Update releases verse ve	Favorite ds Releases
Local directory/Medi	All Product Sectors 👻 Go	Provide Feedback Server Downtime Privacy Policy
Filter releases by:	1 results for ALL	
Family ALL (1) IC615 (1)	Rele_ A Versi_ A Platf_ A Type A Search Search All Platforms V All Types Size(_ A	EOS State
	✓ IC-615_06 IC-615~06 Inx86 Hottrx 4881 Nov 30, 19 No	vy 30, 19 Not installed
	DisplayName: IC-615 06.15.132 RHEL 4(Inx86) Hotfix Release Platform: Inx86 DownloadSize: 4881 MB Description: IC-615 06.15.132 RHEL 4(Inx86) Hotfix Release URL:	
<u>. </u>	Cancel Start Search Again	Next>

Figure A.2: Installing Virtuoso from Installscape Step 2

\$	InstallScape(TM) Classic View	_ = >
InstallScape(TM) Class	ic View cāde	nce
Eile <u>V</u> iew <u>P</u> references <u>H</u> e		nee
View installed Configure releases releases Local directory/Med	Update releases uninstall Search & Install releases unistall Control file install unistall unistall	
۹ [All Product Sectors V Go	edback vntime icy
Filter releases by: Release Name Family		
ALL (1) IC615 (1)	Choose Products Check the products that you want to install and uncheck the products that you don't want to install.	^
	ID Name Size (MB) Version	
	PIC-615 IC615_INX86 7846	
	- D - 111 111:Cadence(K) 1414 111:06.15.132	
	P12141_64bit 12141_64bit:121 376 12141_64bit=06	
	C P12141 12141:Cadence(1594 12141-06.15.13	
	Required Disk Space: 7846MB	
	Choose Install Options	=
	Directories Select Install directory (Available Space: 11375 MB)	
	/usr/local/Cadence/IC6.1 Use defaults Browse	
	Progress meter Total Estimated time Elapsed: Installing: Total Estimated time Remaining (approx): Installing:	
	0% Configuring:	
	0%	-
	<u>Cancel install</u> < <back< td=""><td>Start</td></back<>	Start

Figure A.3: Installing Virtuoso from Installscape Step 3

)	InstallScape(TM) Classic View	_ 0
nstallScape(TM) Clas	ssic View	cādence
Eile ⊻iew Preferences H	elp	
View installed Configure releases releases	Update releases Uninstall releases Uninstall releases Uninstall releases Uninstall releases Uninstall Uninstall Search & Install Control file install Uninstall Control file install Uninstall Control file install Uninstall Control file install	
Local directory/Me	dia instali	
٩	All Product Sectors 💌 Go	Provide Feedback Server Downtime Privacy Policy
Filter releases by:	<u>[1</u>	•
Release Name Family	<pre>// CC-615_06.15.132_Inx86</pre>	
ALL (1)		
IC615 (1)	Check the products that you want to install and uncheck the products that you don't want to	install.
	ID Name Size (MB)	Version
	✓ ■ PIC-615 IC615_Inx86 7846	
	– 🗹 🗋 P111 111:Cadence(R) Desi 1414	111~06.15.132~Inx
	P12141_64bit 12141_64bit: 12141 376	12141_64bit~06.15
	⊷ 🖉 🚍 P12141 12141:Cadence(R) D 1594	12141~06.15.132~I
	Required Disk Space: 7846MB	
	Choose Install Options	
	Directories	
	Select Install directory (Available Space: 11375 MB)	
	/usr/local/Cadence/IC6.1	faults Browse
	Prograss mater	
	Total Estimated time Elapsed: 00:01:16 hh:mm:ss Total Estimated time Remaining (app	prox): 00:03:14 hh:mm:
	Installing: 2395903 K of 8035063 K (129 components of 426 components)	, i i i i i i i i i i i i i i i i i i i
	29%	
	Configuring:	
	0.6	• • • • • • • • • • • • • • • • • • •
		- Church
	<u>Cancelinstan</u> << Back	Start

Figure A.4: Installing Virtuoso from Installscape Step 4

٩	InstallScape(TM) Classic Vie	×
InstallScape(TM) Classic Vie	<i>,</i>	cādence [®]
<u>F</u> ile ⊻iew <u>P</u> references <u>H</u> elp		
View installed Configure Upd releases releases release	te Uninstall Search & Install Control file releases	Local directory/ Media install Downloads Releases
Local directory/Media ins	all	
Q.	All Product Sectors 👻	Provide Feedback
		Mult
Filter releases by: © Release Name O Family AL (3) IC615 (1)	To co media IC 06	mplete installing this release, you must specify the following 1. .15.011 RHEL 4.0(Inx86) Base Release Disk 1 Point [\STALL/IC6.1/IC06.15.011_Inx86.Base/CDROM1] Browse]
	Choose Install Options Directories Select Install directory (Availat Uner Install Carbone IICE 3	X Cancel
	Progress meter Total Estimated time Elapsed: 00:04:48 hhrmm:ss Installing: 7666435 K of 8035063 K (385 compone Configuring:	Total Estimated time Remaining (approx): 00:01:12 hh:mm of 426 components) 95% 0%
	Cancel Install	< <back start<="" th=""></back>

Figure A.5: Installing Virtuoso from Installscape Step 5

•	InstallScape(TM) Classi	c View	_ = ×
InstallScape(TM) Classic Vic	NM/		cādon co [®]
Cadeno	ce_ncvhdl64b10.20-s118lnx86_Config	_ = ×	cauence
Cademan novhd1(64): 10.20-s118: (c) Cogeright NEINERNEL, arc/kague.vhd: NEINERNEL, arc/kague.vhd: NEINERNEL, arc/sparse.vhd: NEINERNEL, arc/sparse.vhd: novhd1(64): 10.20-s118: (c) Cogeright errors: 0, warnings: 0 novhd1(64): 10.20-s118: (c) Cogeright .v/IEEE.vhd.ma.src/shartenel.constarts .v/IEE.vhd.ma.src/shartenel.constarts .v/IEE.vhd.ma.src/shartenel.constarts .v/IEE.vhd.ma.src/shartenel.constarts .v/IEE.vhd.ma.src/shartenel.constarts .v/IEE.vhd.ma.src/shartenel.constarts .v/IEE.vhd.ma.src/shartenel.constarts .v/IEE.vhd.ma.src/shartenel.constarts .v/IEE.vhd.ma.src/shartenel.constarts .v/IEE.vhd.ma.src/shartenel.constarts .v/IEE.vhd.ma.src/shartenel.constarts .v/IEE.vhd.mas.src/shartenel.comstarts .v/IEE.vhd.	re norkhildebild.20-allBinx86_Conng 1985-2012 Cadence Besign Systems, Inc. 1985-2012 Cadence Besign Systems Jund 1985-2012 Cadence Besign System di 1985-2012 Cadence Besign System di 1985-2012 Cadence Besign System di 1985-2012 Cadence Besign System di 1985-2012 Cadence Besign System	32_inx86 typu don't want to install. 7846 gn1414 11-06 64376	Provide Feedback Server Downtime Privacy Policy
Run setenv ISCAPE_SLEEP <n> to change</n>	to <n> seconds</n>	esi 1594 12141~	06.15.132~Inx86
Press Enter to close this window			······
20 19 18 17 16 15 14 13 12 11 10 9 8 7	65430		
		•	
	Choose Install Options	R:	
	-Directories Select Install directory (Available Space: 3115 MB)		
		Use defaults	Browse
	Progress meter Total Estimated time Elapsed: 00:06:15 hh:mm:ss Installing: 8038076 K of 8035063 K (426 components of 42	Total Estimated time Remaining (approx): 00:00 6 components)	0:10 hh:mm:ss
		100%	
	Configuring: Configuring 2 of 9 components	2.20/	
		2.276	
	<u>Cancel Install</u>	< < Back	Start

Figure A.6: Installing Virtuoso from Installscape Step 6

٩	InstallScape(TM) Cla	issic View	_ 🗆 X
InstallScano/TML Classic M	liow .		cādence°
Cade	ence_oaRedist22.41-p025Inx86_Config	_ 0 ×	
Welcome to the OpenAccess 2.2 Config	guration Utility.	i in the second secon	
Important: Before continuing, first libraries. Then make sure that there (kill these processes with the kill	close all applications that use OpenAccess s are no calMTurbOServer processes running command on UNIX/Linux).	nove Favorite nloads Releases	
Press Enter to continue (q to quit):	: 1°H'H^[^H		
This utility lets you specify a diff	Perent OpenAccess installation to use.		Provide Feedback
The current installation is [/usr/lc Do you want to use a different Openf	acal/Cadence/IC5,1/oa_v22,41,025] Access installation (y/n/q) ? [y] y		Server Downtime Privacy Policy
		32_Inx86	
			A
		t you don't want to install.	
		Size (MB)	Version
		7040	-06 15 123-Jpv86
		64 376 121	41 64bit=06 15 13
		esi 1594 121	41~06 15 132~lnx86
	· · · ·		
	Choose Install Options	k	
	Directories		
	Select Install directory (Available Space: 3115 MB)		
		Use defau	ilts Browse
	Progress meter		
	Installing: 8038076 K of 8035063 K (426 components o	f 426 components)):00:10 nn:mm:ss
		100%	
	Configuring: Configuring 3 of 9 components	2.38/	
		33/0	
i _			v
	Cancel Install	< <back< th=""><th>Start</th></back<>	Start

Figure A.7: Installing Virtuoso from Installscape Step 7

٩	InstallScape(TM) C	Classic View				_ = ×
InstallScane/TM) Classic V					cā	d e n c e°
Institlection (TRA) Clearie 2 Geometry (Clearies) (Clea	inne_oaRedist22.41-p025Inx86_Config writion Utility, close all applications that use Qeenécess are no adMIroberve processes unning command on (NIV/Linux),	- ° X	32_Inx	Favorite Favorite Feleases	C ā	dence defeedback er Doxmine acy Pallor
	Choose Install Options Directories Select Install directory (Available Space: 3115 MB) [forr/local/Cadence/IC6] Progress meter Total Estimated time Elapsed: 00.06:15 hh:mm ss Installing: 8038076 K of 8035063 K (426 components Configuring: Configuring 3 of 9 components	Total Es s of 426 components) 100% 33%	_64 esi k	376 1594 Use	12141_64bit=06.15.13 12141_06.15.132_inx86 200_0010_15.132_inx86 edefaults Browse x0: 00:00:10 hh:mm:ss	
	<u>Cancel Install</u>		< < B a	ck		Start

Figure A.8: Installing Virtuoso from Installscape Step 8

٩	InstallScape(TM) Classic View	N			_ = ×
InstallScane(TM) Classic \/ie				c 5 d	0 D C 0°
Cadence	e_lfvhdComp02.85-a002lnx86_Config	×		נמט	ence
Proceeding with configuration of lfvhd	Comp				
Synergy users must install these libra Install Valid library and Sim primitiv Install XL library and Gate primitive Installing Leapfrog Analyzer and Desig Library installation	ries vee?(<cr> = n): ^[\$ (<cr> = n): 1 library</cr></cr>	no nic	ove Favorite oads Releases		
vrelocate 2,85-a002: (c) Copyright 199 vrelocate 2,85-a002: (c) Copyright 199 vrelocate 2,85-a002: (c) Copyright 199	2-1997, Cadence Besign Systems, Inc. 2-1997, Cadence Besign Systems, Inc. 2-1997, Cadence Besign Systems, Inc.	- 1		Provide	Feedback
Installation of Leapfrog Analyzer and .	Design libraries complete			Server I Privacy	Downtime Policy
This window will close in 20 seconds Run setenv ISCAPE_SLEEP <n> to change</n>	to ⊲N> seconds				•
20 19 18 17 16 15 14 13 12 11 10 9 8 7	6	32.	_Inx86		
					^
		ty	ou don't want to install.		
			Size (MB)	Version	
			7846		
		gn	1414	111~06.15.132~Inx86	
		_6	4 376	12141_64bit~06.15.13	
		esi	i 1594	12141~06.15.132~Inx86	
		-			
	Choose Install Options	R.			
	Directories				
	Select Install directory (Available Space: 3115 MB)				
				Use defaults Browse	
	Progress meter				
	Total Estimated time Elapsed: 00:06:15 hh:mm:ss	Total Estin	mated time Remaining (app	prox): 00:00:10 hh:mm:ss	
	Installing: 8038076 K of 8035063 K (426 components of 426 com	nponents)			
	Configuring: Configuring 4 of 9 components	100%			
	comparing. Comparing 4 or 5 components	44%			
	<u>Cancel Install</u>	<	<back< td=""><td></td><td>Start</td></back<>		Start

Figure A.9: Installing Virtuoso from Installscape Step 9

٩	InstallScape(TM) Cla	ssic View				_ = ×
InstallScane/TM) Classic V	iow .				cād	e n c e°
Cad This window will close in 20 seconds Run setery ISCAPE_SLEEP <n> to chang Press Enter to close this window 20 19 18 17 16 15 14 13 12 11</n>	erce_CSRCOIndep07.25-p034_Config	_ D X	nove nload	Favorite s Releases		
			32_Inx	86	Providi Server Privacy	EFeedback Downtime Policy
		1	t you	don't want to install. Size (MB)	Version	
		ľ	gn	7846 1414 376	111~06.15.132~Inx86	
			esi	1594	12141_04.15.132~Inx86	
	Choose Install Options Directories Select Install directory (Available Space: 3115 MB) [/usr/local/Cadence/IC6-1		j	Use	defaults Browse	
	Progress meter Total Estimated time Elapsed: 00:06:15 hlt.mm:ss Installing: 8038076 K of 8035063 K (426 components of Configuring: Configuring 5 of 9 components	Total Es f 426 components) 100% 55%	itimate	d time Remaining (approx): 00:00:10 hh:mm:ss	
PJ ()	<u>Cancel Install</u>		< < B a	ck		Start

Figure A.10: Installing Virtuoso from Installscape Step 10

٩	InstallScape(TM) Classic View					_ = ×
InstallScano(TM) Classic View					c.5.d	0.0.0
Caden	ce_deft01.03-s044lnx86_Config	. 🗆 X			ιαυ	ence
This window will close in 20 seconds Ran seterw ISDAPE_SLEEP (n) to change to Press Enter to close this window 20 19 18 17 16 15 14 13 []	QD seconds		nove nove	Favorite Releases		
					Provid Server Privac	e Feedback Downtime Policy
			32_In	x86		,
			t you	don't want to install.		
				Size (MB)	Version	
				7846	Ē	
			gn	1414	111~06.15.132~Inx86	
			esi	1594	12141~06 15 132~lnx86	
	Choose Install Options))			
	Directories					
	Select Install directory (Available Space: 3115 MB)					
				Use	defaults Browse	
	Progress meter Total Estimated time Elapsed: 00:06:15 hh:mm:ss Installing: 8038076 K of 8035063 K (426 components of 426 compo	Total E onents)	stimat	ed time Remaining (approx	(): 00:00:10 hh:mm:ss	
	Configuring: Configuring 7 of 9 components					
	7	77%				
	<u>Cancel Install</u>		< < B	ack		Start

Figure A.11: Installing Virtuoso from Installscape Step 11

٩	InstallScape(TM) Class	ic View			_ = ×
InstallScano/TM) Classic Via				cā	d e n c e°
The hierarchy has already been configur This window will close in 20 seconds Rum setern JSCHFE_SLEEP (or to change 1 Press Enter to close this window 20 19 18 17 16 15 14 13 12 11 10 9	ed to use the odsLib plugin. a OD seconds		ve Favorite ads Releases		
				Pro Sen Priv	vide Feedback ver Downtime racy Policy
		52 t y	ou don't want to install.		
			Size (MB)	Version	_
		gn	1414	111~06.15.132~Inx86	Ξ
		_64	4 376	12141_64bit~06.15.13	
		esi	1594	12141~06.15.132~Inx86	-
	Choose Install Options Directories Select Install directory (Available Space: 3115 MB)	k		a defaulte - Resure	=
	Progress meter	Total Estin	nated time Remaining (appr	av): 00:00:10 bhimmiss	
	Installing: 8038076 K of 8035063 K (426 components of 4 Configuring: Configuring 8 of 9 components	26 components) 100%	awea arrie Kernanning (appri	55, 55, 55, 55, 10 million, 55	
		88%			
	<u>Cancel Install</u>	<-	<back< th=""><th></th><th>Start</th></back<>		Start

Figure A.12: Installing Virtuoso from Installscape Step $12\,$
٩	InstallScape(TM) Class	sic View				_ = ×
InstallScano/TM) Classic M	liow				h E o	οποο
c	adence_ihdl06.15-s089lnx86_Config	×			ιαυ	ence
Info: Starting NC Cleanup NC-Clean up process completed succes This window will close in 20 second Rum setern ISOAPE_SLEP <a box="" chang<br="">Press Enter to close this window 20 19 18 17 16 15 14 13 12 11 10 9 8	ssfully. Check the ./CleanNC.log file for more details. 5 ge to 400 seconds 376543[nove nove	Favorite Releases		
			32_Inx	86	Provide Server Privacy	e Feedback Downtime Policy
			t you a	don't want to install.		
				Size (MB)	Version	
			an	1414	111-06-15-122-lbx96	
		-	gn	276	111~00.15.152~INX80	
			ori	1504	12141_045(=00.15.15	
			esi	1.1.0		
	Choose Install Options Directories Select Install directory (Available Space: 3115 MB) [/usr/local/Cadence/IC6.1 Progress meter Total Estimated time Elapsed: 00.06.15 hh:mm:ss Installing: 8038076 K of 8035063 K (426 components of - Configuring: Configuring 9 of 9 components	Total Es 426 components) 100% 100%	timate	Use Use	defaults Browse): 00.00:10 hh:mm:ss	=
	Cancel Install		< < B a.	ck		Start

Figure A.13: Installing Virtuoso from Installscape Step 13

InstallScape(TM) Classic View Cādence Ele Yiew Perenoss Belp View installe Configure View Bases Perenose Periods Per	٢	InstallScape(TM) Classic View		_ 🗆 ×
	InstallScape(TM) Classic View				cādence
Image: Search & Install Image: Search & Install Image: Search & Install Image: Search & Install Image: Search & Install Image: Search & Install Image: Search & Install Image: Search & Install Image: Search & Install Image: Search & Install Image: Search & Install Image: Search & Install Image: Search & Install Image: Search & Install & Install Image: Search & Install &	Eile <u>V</u> iew Preferences <u>H</u> elp				
Local directory/Media install All Product Sectors Go Filter releases by: Filter rele	View installed Configure Update releases releases	Uninstall Search & Install Control fi	le Local directory/ Media install Downloads	Favorite Releases	
Filter releases by: ID Version Install Status Configure Status Filter releases by: IC-615.06.15.132_Inv86 06.15.132 N/A purcess purcess Filter releases by: IC-615.06.15.132_Inv86 06.15.132 N/A purcess purcess	Local directory/Media install				
ID Version Download Status Install Status Configure Status © Retase Name IC-615_06.15_132_inx86 06.15_132 N/A Success Success C615_06.15_132_inx86 06.15_132 N/A Success Success Success C615_01 ID ID Version ID N/A Success Success	<u>م</u> ا	All Product Sectors 🔻 Go			<u>Provide Feedback</u> <u>Server Downtime</u> <u>Privacy Policy</u>
• Retase Name C-615.06.15.132_Inv86 06.15.132 MA purcess purcess PALLIN C615 (1) Filler	Filter releases by:	ID Version	Download Status	Install Status	Configure Status
K4. (4) IC6 15 (1)	Release Name IC-615_06.1	5.132_Inx86 06.15.132	N/A	Success	Success
Nit ccRade Done			- eBat	•	Down

Figure A.14: Installing Virtuoso from Installscape Step 14

A.2.2 Installing MMSIM (Spectre/SpectreRF/HSpice)

- 1. Now that you have installed virtuoso in order to actually use the HSPICE or Spectre simulation engines you need to install the MM-SIM package.
- 2. Download the MMSIM installation files from your ftp server and keep the tar files in the Downloads folder. You will have 3 files for Base version and 3 files for the Hotfix version. Extract the files one by one into the /home/EEAPPS/CADENCE_INSTALL. Make sure that while you untar your files you 'untar' each 'Base' file into the same folder and each 'Hotfix' file into the same folder so at the end of the whole process you will have two folders inside the 'CADENCE_INSTALL' folder named MMSIM11.10.214_lnx86.Base and MMSIM11.10.617_lnx86.Hotfix [14]. Although all the entire installation will be performed from the Hotfix files it is very important to also have the Base files extracted as a path to them will be needed during the installation setup just like you did during Virtuoso installation.
- 3. Open up the terminal window and create a new directory inside the CADENCE_INSTALL folder by typing in *mkdir -p /home/EEAAPS/-CADENCE_INSTALL/MMSIM11.1/*. Now move the extracted Base and Hotfix folders to the 'MMSIM11.1' folder you just created.
- 4. In the terminal window browse to the following folder: cd /home/EEAPPS/CADENCE_INSTALL/ MMSIM11.1/MMSIM11.10.617_lnx86.Hotfix/CDROM1 and then type in sh SETUP.SH to start the installation process.
- 5. You will now see the following instructions so once again follow the steps indicated below very carefully:
 - (a) Specify path of install directory [OR type [RETURN] to exit]: /home/EEAPPS/MMSIM11.1
 - (b) Do you have InstallScape for lnx86 platform installed somewhere [y/n]? ${\bf y}$
 - (c) Type the path to InstallScape installation directory [(q to exit)]: /home/Cadence/InstallScape

- 6. Now a window like Figure A.15 will pop up so follow the instructions shown in it to browse into the correct folder that contains the installation files.
- 7. From this point onwards follow the instructions shown in Figures A.16 through A.22 very carefully to complete the installation process for Virtuoso. Make sure you do exactly as shown in these figures to ensure the software gets installed properly.
- 8. Once you reach the last step as shown in Figure A.22 hit 'Done'.

٩	InstallScape(TM) Classic View	_ 🗆 ×
InstallScape(TM) Classic Vi	iew	cādence
<u>File View P</u> references <u>H</u> elp		
View installed Configure Up releases releases rel Local directory/Media in	odate Uninstall Search & Install Control file releases releases releases stall	
۹	All Product Sectors 👻 Go	<u>Provide Feedback</u> <u>Server Downtime</u> <u>Privacy Policy</u>
Filter releases by: @ Release Name > Family **No matching results found**	Select location of media /usr/local/CADENCE.INSTALL/MMSIMI1.1/MMSIMI1.10.617.Inx86.Hotfix/CDROM1 Type in location of media directory or click on Browse to select a media directory	▼ Browse
	Exit	Continue>

Figure A.15: Installing MMSIM from Installscape Step 1

٩	InstallScape(R) Classic View	_ = ×
InstallScape(R) Classic Vi	ew	cādence°
Eile View Preferences Help		
View installed Configure releases releases ro Local directory/Media i	Dedate Uninstall Search & Install Control file Install Control file Install Install Control file Install	
۹. [All Product Sectors 👻 Go	<u>Provide Feedback</u> Server Downtime Privacy Policy
Filter releases by:	1 results for ALL	
Release Name Eamily	Release A Version A Platform A Type A	
ALL (1)	Search Search All Platforms V All Types V SizeOMB) A Release Date EOS Date A	State 🛋
MMSIM111 (1)	MMSIM_11.10.6 MMSIM~11.10.6 Inx86 Hotfix 1738 Nov 30, 1971 Nov 30, 1971	Not Installed
	Refease Details	
	Unsplaywalle: Wmosim 11.001/ HFEL SUITAGD HUUK Helease Platform: INS6 DownloadSize: 1738 MB Description: MMSIM 11.10.617 RHEL 5(Inx86) Hotfix Release URL:	
	Cancel Start Search Again	Next>

Figure A.16: Installing MMSIM from Installscape Step 2

•	InstallScape()	R) Classic View		_			
InstallScape(R) Classic Vie	w			cādence			
Eile ⊻iew <u>P</u> references <u>H</u> elp				cadence			
View installed Configure releases configure tocal directory/Media in	odate Uninstall Search & Install Control fill releases Install All Product Sectors	e Local directory/ Media install Downloads	Favorite Releases	Provide Feedback Server Downtine			
				Privacy Policy			
Filter releases by:	(MMSIM_11.10.617_Inx8	16)			
ALL (1) MMSIM111 (1)	MMSIM 11.10.617 RHEL 5(Inx86) Hotfix Release Choose Products						
	Check the products that you want to install and uncheck the products that you don't want to install.						
		Name	Size (MB)	Version			
		MMSIM111_INX86	2528				
	F 192301	32501:VIRtuoso(R) Spectr	1127	32501~11.10.617~Imx86			
		33400:Virtuoso Oltrasim	1929	33400~11.10.617~INX86			
		33580:VIrtuoso(R) Keixp	2435	33580~11.10.617~INX86			
	Required Disk Space: 2528MB	CARACIU. ALC.					
			6				
	Choose Install Options	24	,				
	Select Install directory (Available Space: 2969	MB)					
	/usr/local/Cadence/MMSIM11.1		Use	defaults Browse			
	Progress meter Total Estimated time Elapsed: Installing:	Total Estimate	d time Remaining (approx)				
	Configuring	0%					
	company,	0%					
	Cancel Install	< <bac< th=""><th>k</th><th>Start</th></bac<>	k	Start			

Figure A.17: Installing MMSIM from Installscape Step 3

InstallsCape(R) Classic View Piere view installed Configure releases rele	٩				InstallScape(R) Classic View	_ = ×
Elle Lylew Perferences: Help View installed Configure Update United to the finance of the f	InstallScape	(R) Classic V	′iew			cādence°
View installed Configure leases Update releases Update releases Dear of a long o	<u>F</u> ile <u>V</u> iew <u>P</u> r	eferences <u>H</u> elp				
Local directory/Media install Provide is estaback Served is install Provide is estaback Served is install Provide is estaback Filter releases by: Interns & Conditions Terms & Conditions <p< th=""><th>View installed releases</th><th>Configure releases</th><th>Update Uninstall releases releases</th><th>Searc</th><th>h & Install Control file Local directory/ Remove Favorite Jeases install Media install</th><th></th></p<>	View installed releases	Configure releases	Update Uninstall releases releases	Searc	h & Install Control file Local directory/ Remove Favorite Jeases install Media install	
All Product Sectors Go Terms & Conditions Prave Edeases hy: Inter releases hy: Inte	Local direc	tory/Media	install			
Itter releases by: Iterms & Conditions Itter releases Name Iterms & Conditions Release Name Iterms & Conditions Automatic Conductions Iterms & Condititers Automatic Condecons	٩		All Produ	ct Secto	rs 💌 Co	Provide Feedback Server Downtime Privacy Policy
 Relaxes Name Family MMSIM 11.1.0.617 Rf Choose Froducts Check the produ PHAISIN P3256 P334 P3257 P3257 Required Disks Choose Install on Productors Choose Insta	Filter release	s by:	1	U Torr	terms & Conditions	×
Start	(Redae Raf (Radie AL, 0) MMSIM111 (1)		MMSIM 11.10.617 RH Choose Products Check the prod PMSIM P 1925C P 1925	•	Please read the following terms and conditions (Terms and Conditions') carefully. This product, any additional related products or deliverables, and any copies confidential and proprietary property of Cadence Design Systems (including any of its subsidiaries, 'Cadence') or tindri paries form whom Cadence has obtained rights and are protected by United States copyright laws, other local laws and intermational tready provisions. You may only use the product(s) under a valid written license agreement with Cadence and subject to the terms to obtained rights and are protected by United States copyright laws, other local laws and intermational tready provisions. You may only use the product(s) under a valid written license agreement with Cadence and subject to the terms any product or portion tiernes agreement. You further agree to take all reasonable steps to protect the product(s) from unauthorized reproduction, publication, use reasonable mechanism in place to strong than there agree to adrise all reasonable mechanism in place to strong than and transmit non-proprietary data on unauthorized modifications to the product(s). No such data will be collected or transmitted unless unauthorized modifications to a product or unauthorized by demessive that you are only using authorized modifications to the product(s), we under agree to take all reasonable being of the license server or product(s) are abuggeted. by deminabing the product(s), we underbarge that you have ther the Cadence Proxy Policy, which is incorporated to these Terms and Conditions by reference, may be found at the Ty/lowx cadence com/pages/page to the use of such technology and transmission of such non-proprietary data by Cadence. Terms and Conditions by reference, may be found at the product(s) and direcontinue any use.	* ************************************

Figure A.18: Installing MMSIM from Installscape Step 4

InstallScape(R) Classic View Câdence* Initial View Deferences Help View Install Configure Update releases Unitial Search & Install Control file Local directory/ Remove Favorite releases Cocal directory/Media Install Control file Local directory/ Media Install Cocal directory/Media Install Method install Product Sectors Colspan="2">Colspan="2">Colspan="2">Colspan="2">Control file Local directory/ Media Install Method install Init init init init init init init init	٩	Install	IScape(R) Classic View	_ 0 ×
Ele Yew Peferences Help View installed Configure Update Units all releases Units all View Peferences Help Local directory/Media Install Control File releases Configure View Peferences Help Control File releases Configure View Perences Help Control File Remove Pavorite Downloads Releases Products Rectances Product Sectors Configure View Perence Perence Products Releases Product Sectors Configure View Perence Product Perence Product Sectors Configure View Perence Product Perence Product Perence Product Sectors Configure View Perence Product Per	InstallScape(R) Classic View			cādence°
View installed Configure releases Update releases Uninstall velocities Control file local directory/ Media install Control file local directory / Media install Control file local directory / Media install Control file local directory / Media install ErrodeE Feedback. Server Downloads Releases Local directory / Media install Interview Product Sectors Colspan="2">Control file local directory / Media install Produe Feedback. Sector Committee Product Sectors Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Release Name Image: Name Image: Release Name Image: Name Image: Name Image: Release Name Image: Release Name Image: Name	Eile View Preferences Help			tudente
Filter releases by: Image: Second	View installed Configure releases releases Local directory/Media install	tall Search & Install Cor releases i Product Sectors V Co	Introl file Install Development Downloads Releases	Provide Feedback Server Downtime
Filter releases by: 				Privacy Policy
Instributed // Cadence // MI/SM Browse Progress meter Total Estimated time Blapsed: Total Estimated time Blapsed: OK Installing: 197/130 K of 258(Configuring: 0%	Filter refeases by: © Refeases Name ○ Family All (1) All (1)	17 RHEL SOnx lucts products that you 10 451M 10 451M 10 452501 17 233500 233520 233520 233520 233520 233520 233520 233520 233520 233520 233520 23520	X To complete installing this release, you must specify the following media: MMSIM 11.10.214 RHEL 5(Inx86) Base Release Disk 1 Mount Points (SIM11 1/MMSIM11 10.214_Inx86 Ease/CDROM1) Rrowse OK Cancel 76% 0%	rsion • 0.617-inx86 • 0.617-inx86 • 0.617-inx86 • Browse •
Cancelinstall <<8ack Start	<u>، ا</u>	Cancel Install	< < Back	Start

Figure A.19: Installing MMSIM from Installscape Step 5

٩	InstallScape(R) Classic View				_ = ×
InstallScane/P) Classic View				cāde	nce
Cadence_oaRedist22.4	1-p025lnx86_Config _ C	×		Cuuc	ince
		_			
Welcowe to the OpenAccess 2.2 Configuration Utility.		0			
Important: Before continuing, first close all application libraries. Then make sure that there are no calMTurboServ (kill these processes with the kill command on UNIX/Linux	s that use OpenAccess er processes running).	nove iloa	e Favorite ds Releases		
Press Enter to continue (q to quit):					
This utility lets you specify a different OpenAccess inst	allation to use.			Provide F	eedback
The current installation is [/usr/local/Cadence/HMSIM11.1 Do you want to use a different OpenAccess installation (y	/oa_v22.41.025] /n/q) ? [y] n[]			Server Do Privacy Pr	wntime plicy
		.7_lr	1×86		
		t you	ı don't want to install.		
			Size (MB)	Version	
		ect	. 1127	32501~11.10.617~Inx86	
		5im	. 1929	33400~11.10.617~Inx86	
		IX	2435	33580~11.10.617~Inx86	
Choose Install	Options		k		
Directories					
Select Inst	all directory (Available Space: 362 MB)				
/usr/local			Use	e defaults Browse	
Progress m	ter				
Total Estimat	ed time Elapsed: 00:04:09 hh:mm:ss To	ital Estima	ted time Remaining (appro	x): 00:00:37 hh:mm:ss	
Installing: 2	89337 K of 2588745 K (63 components of 63 components 1009)			
Configuring:	Configuring 1 of 1 components				
	1009	6			
Can	<u>sel Install</u>	< < E	Back	[Start

Figure A.20: Installing MMSIM from Installscape Step 6

	InstallScape(R) Class	sic View			
actallScape/D) Classic Vi	(a).W			cād	епсе
Ca	dence_oaRedist22.41-p025lnx86_Config	_ = ×			CIICC
Welcome to the OpenAccess 2.2 Cont	figuration Utility.	6			
Important: Before continuing, firs libraries. Then make sure that the (kill these processes with the ki	st close all applications that use OpenAccess ere are no oaUMTurboServer processes running 11 command on UNIX/Linux).	nove F 1 loads R	avorite eleases		
Press Enter to continue (q to qui*	t):				
This utility lets you specify a d	ifferent OpenAccess installation to use.			Provid	e Feedback
The current installation is [/usr/ Do you want to use a different Opr	/local/Cadence/HHSIM11.1/oa_v22.41.025] anRccess installation (y/n/q) ? [y] n			Server Privacy	Downtime V Policy
This window will close in 20 secon Run setenv ISCAPE_SLEEP <n> to ch/</n>	nds ange to ⊲N⊳ seconds				Þ
20 19 18 17 16 15 14 13 12 11 10	I	.7_ln×86			
					^
		t you don'	t want to install.		
			Size (MB)	Version	
		252	8		
		ect 112	7	32501~11.10.617~Inx86	
		5im 192	9	33400~11.10.617~Inx86	
		DX 243	5	33580~11.10.617~Inx86	
	3 1 1 1 1		-		
	Choose Install Options	k			
	Directories				
	Select Install directory (Available Space: 362 MB)				
	/usr/local/Cadence/MMSIM11.1		Use	e defaults Browse	
	Progress meter				
	Total Estimated time Elapsed: 00:04:09 hh:mm:ss Installing: 2589337 K of 2588745 K (63 components of 6	Total Estimated tir 3 components)	ne Remaining (appro	x): 00:00:37 hh:mm:ss	
		100%			
	Configuring: Configuring 1 of 1 components	100%			
		100%			
					-
			1		_ e

Figure A.21: Installing MMSIM from Installscape Step 7

٩		InstallScape(F	t) Classic View		_ = ×
InstallScape(R) Classic Vi	iew				cādence°
<u>F</u> ile <u>V</u> iew <u>P</u> references <u>H</u> elp					
View installed Configure releases releases	Update Uninstall Sea	arch & Install Control file releases install	Local directory/ Media install	Favorite Releases	
Local directory/Media	Install				
م [All Product Ser	ctors 🔻 Go			Provide Feedback Server Downtime Privacy Policy
Filter releases by:	ID	Version	Download Status	Install Status	Configure Status
AL (1) MMSIM 111 (1)				•	
	<u>Exit</u>		< <back< td=""><td>]</td><td>Done</td></back<>]	Done

Figure A.22: Installing MMSIM from Installscape Step 8

- 9. Now that you have installed both Virtuoso and MMSIM the most critical step is to configure the environment variables correctly. In order to do so you will need to change your OS's shell to bash. To figure out the current shell of your OS open up a terminal and type in echo \$SHELL. To change the shell to bash if it is not set by default type in chsh -s /bin/bash. If you actually were successful in changing the shell type in echo \$SHELL, and you should get /bin/bash as an output.
- 10. Open up your current *bash* file by typing *gedit* .*bashrc* & in the terminal window. Now replace the text with that of Section 2.3. Save the updated file and close it.
- 11. In the terminal window type in **source .bashrc** to update your *bash* settings.

A.2.3 Installing a PDK

- 1. Download the PDK from your foundry vendor and extract the files in a new directory called PDK under the path: */home/EEAPPS*.
- 2. Create a working directory in your home folder as this will be the directory where you will launch Virtuoso from and will store all your files. For the purposes of this tutorial call your work directory *ckt180* under your 'Documents' folder.
- 3. Copy the *cds.lib* file from the PDK you installed above and open it up in a text editor.
- 4. Make sure your 'cds.lib' has the following items before you launch virtuoso. We have to do this to include the in-built libraries that come with the Virtuoso software.
 - (a) SOFTINCLUDE /home/EEAPPS/IC615/share/cdssetup/cds.lib
 - (b) SOFTINCLUDE /home/EEAPPS/IUS08.20.015/tools/inca/files/cds.lib
 - (c) SOFTINCLUDE /home/EEAPPS/TSMC018/cds.lib
 - (d) DEFINE and Lib \$CDSHOME/tools/dfII/samples/artist/and Lib
- 5. At this point Virtuoso is ready for launch so browse to your work directory and type **virtuoso &**. See Figure A.23.

A.2.4 Remote Connections Setup

In order to remotely login to the Server from your machine follow the instructions provided below:

1. Windows OS Users:

- (a) Install the SSH client MobaXterm[http://mobaxterm.mobatek. net/download-home-edition.html] depending on your preference.
- (b) Install Xming X Server [http://sourceforge.net/projects/ xming/] for Windows to allow X-forwarding during the SSH session. Also, install Xming-fonts from [http://sourceforge.

net/projects/xming/files/Xming-fonts/]

<u>Note:</u> Without installing Xming you will not be able to open Virtuoso or for that matter any application with a GUI.

(c) Launch your SSH client, type ssh -X username@natcsi.ece.illinois.edu, hit 'Enter'. You will prompted to type in a password so type it in and again hit 'Enter'. Now you can follow the steps outlined in Figure A.23.

2. Mac OSX Users:

- (a) Install XQuartz 2.7.5 for Mac OSX if you are using OSX Mountain Lion or later. If you have an older OS then you will already have X11 pre-installed in your system. Check your 'System Preferences' to check whether X11 is turned on.
 <u>Note:</u> Without installing XQuartz or enabling X11 (depending upon your OSX version) you will not be able to open Virtuoso or for that matter any application with a GUI.
- (b) Launch your SSH client and type ssh -X username@natcsi.ece.illinois.edu, hit 'Enter'. You will prompted to type in a password so type it in and again hit 'Enter'. Now you can follow the steps outlined in Figure A.23.

3. Linux OS Users:

(a) Launch Terminal and type ssh -X username@natcsi.ece.illinois.edu, hit 'Enter'. You will prompted to type in a password so type it in and again hit 'Enter'. Now you can follow the steps outlined in Figure A.23.



Figure A.23: Launch Instructions for Virtuoso

A.2.5 Configuring Bash Environment

```
# .bashrc
# Source global definitions
if [ -f /etc/bashrc ]; then
       . /etc/bashrc
fi
# User specific aliases and function
alias mat='cd /home/rishi/matlab; matlab &'
alias cscope='/home/EEAPPS/Cscope/ai_bin/cscope'
SYNOPSYS_HOME=/home/EEAPPS
HSP_HOME=$SYNOPSYS_HOME/HSPICE
SCL_HOME=$SYNOPSYS_HOME/SCL
HSP_BIN=$HSP_HOME/hspice/bin
SCL_BIN=$SCL_HOME/linux/bin
export LM_LICENSE_FILE=/home/EEAPPS/HSPICE/linmac.dat
export PATH=${HSP_HOME}/hspice/bin:$PATH
export PATH=/home/EEAPPS/Cscope/ai_bin/:$PATH
export MMSIMROOT=/home/EEAPPS/MMSIM
export OAHOME=/home/EEAPPS/IC615/oa
export CDSHOME=/home/EEAPPS/IC615
export CDSDIR=/home/EEAPPS/IC615
export CDS_ROOT=/home/EEAPPS/IC615
export CDS_INST_DIR=//home/EEAPPS/IC615
export DD_DONT_DO_OS_LOCKS=SET
export CDS_LIC_FILE=5280@cadence.webstore.illinois.edu
export CDS_Netlisting_Mode=''Analog"
export_PATH=${CDS_INST_DIR}/tools/bin:$PATH
export_PATH=${CDS_INST_DIR}/tools/dfII/bin:$PATH
export_PATH=${CDS_INST_DIR}/tools/plot/bin:$PATH
export_PATH=${CDS_INST_DIR}/tools/dracula/bin:$PATH
export_PATH=${CDS_ROOT}/tools/bin:$PATH
export_PATH=${CDS_ROOT}/tools/dfII/bin:$PATH
export_PATH=${CDS_ROOT}/tools/dracula/bin:$PATH
```

```
export_PATH=${CDS_ROOT}/tools/plot/bin:$PATH
export_PATH=${CDS_ROOT}/tools/iccraft/bin:$PATH
export_PATH=/home/EEAPPS/InstallScape/iscape/bin:$PATH
export_PATH=${MMSIM_ROOT}/tools/dfII/bin:$PATH
export_PATH=${MMSIM_ROOT}/tools/spectre/bin:$PATH
export_PATH=${MMSIM_ROOT}/tools/ultrasim/bin:$PATH
export_PATH=${MMSIM.ROOT}/tools/bin:$PATH
export_CDS_AUTO_64BIT=ALL
export_CDS_LOAD_ENV=CSF
export_EDITOR=/usr/bin/gedit
export_CADENCE_CURR_IUS=$SYNOPSYS_HOME/IUS08.20.015
export_PATH=${CADENCE_CURR_IUS}/tools.lnx86/bin:$PATH
export_PATH=${CADENCE_CURR_IUS}/tools/bin:$PATH
export_PATH=${CADENCE_CURR_IUS}/bin:$PATH
export_PATH=/home/MATLAB/R2013b/bin:$PATH
```

A.2.6 Creating a Library in Cadence Virtuoso

 The first task after launching Virtuoso is to organize all your designs into appropriate libraries. To view all the libraries in the current work directory click on *Tools* → *LibraryManager* as outlined in Figure A.24, and the Library Manager window will pop up as shown in Figure A.25.

<u>Note:</u> If you want to manually add a library that you copied from an external source into your Cadence work directory you would need to edit the cds.lib file found in your work directory folder by opening it in a text-editor.



Figure A.24: Launch Instructions for Library Manager

• • • • X	Library Manager: WorkArea: /hom	ne/rishi/ThesisSample	
<u>F</u> ile <u>E</u> dit <u>V</u> iew <u>D</u> esign Manager <u>I</u>	lelp		cādenco
Show Categories Show F Library ahdiLib ambit analogLib basic cds_assertions cds_assertions cds_spicelib connectLib ieee ncinternal nemodels ncutlis sdlib stid synoosvy	Cell	View	
Messages	ibManager.log".		
Created new library "TestLib" at /hor	e/rishi/ThesisSample/TestLib.		
elete			

Figure A.25: Library Manager Window

2. To create a new library click on $File \rightarrow New \rightarrow Library$ and name the library as TestLib as highlighted in Figure A.26. After creating the new library you need to specify the Technology File to be used in your respective PDK. In our case we will 'Attach an existing technology library', specifically the 'tsmc18rf' which corresponds to 180nm CMOS process. Figure A.27 shows the steps involved in attaching the appropriate technology file to a new library.

Ele Edit View Design Manager Help Cão	ence
New 🔪 🖬 Library	
Dpm. CH-0 Cell View. Cellence Open (Base-Only). CH-R Cellence Cellence ra Open Witty Cellence Cellence View	
Load Defaults	
Ogen Shell Window Ctrl+P	
Egt CM-X CONFICUE CONFIC	
<pre>c Messages Log file is "htme/Itslu/ThesisSample/IbManager.log"</pre>	



00	X	New Library			
Library —					
Name	TestLib				
Directory		:Sample/	(† E 💣	## ##	
i cds.lik cds.lik libMau libMau) nager.log nager.log.cdslck				
File <u>t</u> ype:	Directories			•	
Design Manager					
● Use N ○ Use N	IONE Io DM				
ОК	Apply	Cance	D	Help	

(b) New Library Name

Figure A.26: Steps to Create New Library

O O X Technology File for New Library	😑 🔿 🔿 🖹 Attach Library to Technology Library		
Technology File for library "TestLib"	New Library	TestLib	
You can: Compile an ASCII technology file Reference existing technology libraries Attach to an existing technology library Do not need process information	Technology Library	amaloglib basic cdsDeffechLib tsmc18ff	
OK Cancel Help		OK Cancel Apply Help	
(a)		(b)	

Figure A.27: Attaching Tech File

Refer to Chapters 6 and 7 for full behavioral/circuit-level simulation guide.

A.3 Common Troubleshooting Tips

Some of the most commonly recurring errors are discussed below along with the possible solution to resolve them:

- 1. Often when working on a common server machine with multiple users logging onto the same account, the cadence filesystems gets "locked" preventing any edit operations on any of the files associated with the given user. The error will resemble the following statement in the terminal window if such an event occurs: *WARNING* file /home-/rishi/CDS.log File is already locked by some other process. In order to fix this problem navigate to your home directory. Note that the home directory is not the directory from which virtuoso is launched, instead it is its the parent directory. Delete any of the files of the form 'CDS.log', 'CDS.log.1' and 'CDS.log.cdslck'. After deleting these files refresh Virtuoso by navigating to the Cadence Virtuoso 'Log' window and clicking on $File \to Refresh$.
- 2. When sharing libraries among users within a server make sure you add the library name in the 'cds.lib' file contained in your cadence launch directory. Additionally, make sure while copying the files, the destination user has write/edit privileges as lack of the write permission will limit edit capabilities within virtuoso.
- 3. During simulation of complex circuits the Cadence simulation folder gets full and often causes the entire machine to hang. In order to prevent this from happening periodically, delete the contents of the 'simulation' folder found within your Cadence launch directory.
- 4. Always save the one functional simulation setting by navigating to the $Session \rightarrow SaveState$ window within the ADE window. Make sure you select 'Cellview' as this would also save the simulation plots from the last simulation, thereby saving some simulation time for future use.

REFERENCES

- S. Palermo, CMOS Nanoelectronics Analog and RF VLSI Circuits, Chapter 9. New York City, N.Y.: McGraw-Hill, 2011.
- [2] E. Alon, "High-speed electrical interface circuit design: Lecture 1," 2011. [Online]. Available: http://bwrcs.eecs.berkeley.edu/Classes/ icdesign/ee290c_s11/lectures/Lecture01_Intro_2up.pdf
- [3] V. Stojanović, "Channel-limited high-speed links: Modeling, analysis and design," Ph.D. dissertation, Stanford University, Palo Alto, 2004. [Online]. Available: http://chipgen.stanford.edu/papers/vs_thesis.pdf
- [4] K. Kundert and O. Zinke, Designer's Guide to Verilog-AMS, Chapter 3. Boston, M.A.: Kluwer-Academic Publishers, 2004.
- [5] D. Friedman, "International solid-state circuits conference trends 2013," 2013. [Online]. Available: http://isscc.org/doc/2013/2013_Trends.pdf
- [6] J. С. "Mutli-gigabit serdes: The Chen, cornerinterconnects," stone of high serial 2011. [Onspeed line]. Available: http://www.design-reuse.com/articles/10541/ multi-gigabit-serdes-the-cornerstone-of-high-speed-serial-interconnects. html
- M. Assaad, "Design and modelling of clock and data recovery integrated circuit in 130 nm cmos technology for 10 gb/s serial data communications," Ph.D. dissertation, Univ. of Glasgow, Glasgow, 2009.
 [Online]. Available: theses.gla.ac.uk/707/1/2009assaadphd.pdf
- [8] P. Hanumolu et al., "Analysis of charge-pump phase-locked loops," *IEEE Transactions on Circuits and Systems-I*, vol. 51, no. 9, pp. 1665– 1674, 2004.
- U.Ku-Moon. P.K. Hanumolu, "Effect of power supply noise on ring osc phase noise," 2004. [Online]. Available: http://web.engr.oregonstate. edu/~moon/research/files/newcas04_supply.pdf

- [10] M. Mansuri, "Low-power low-jitter on-chip clock generation," Ph.D. dissertation, Univ. of California, Los-Angeles, 2003. [Online]. Available: http://www.ece.tamu.edu/~spalermo/ecen689/ pll_thesis_mansuri_ucla_2003.pdf
- [11] U.Ku-Moon. P.K. Hanumolu, G.Y. Wei and K. Mayaram, "Digitallyenhanced phase-locking circuits," in *Proc. IEEE Custom Integrated Circuits Conference* '07), San Jose, USA, Sep. 2007, pp. 361–368.
- [12] V. Kratyuk et al., "A design procedure for all-digital phase-locked loops based on a charge-pump phase-locked-loop analogy," *IEEE Transactions* on Circuits and Systems-II, vol. 54, no. 3, pp. 247–251, 2007.
- [13] J. Fan et al., "Signal integrity design for high-speed digital circuits:progress and directions," *IEEE Transactions on Electromagnetic Compatibility*, vol. 52, no. 2, pp. 392–400, 2010.
- [14] R. Helinski, "Installing cadence ic 6.1," 2010. [Online]. Available: http://www.ece.unm.edu/~jimp/vlsiII/cadence_install/ installing_cadence.pdf
- [15] B. Razavi, Monolithic Phase-Locked Loops and Clock Recovery Circuits, Chapter 1. Piscataway, N.J.: IEEE Press, 1996.